RECLANATION Managing Water in the West



Draft Resource Management Plan: Clark Canyon Reservoir and Barretts Diversion Dam



U.S. Department of the Interior Bureau of Reclamation Great Plains Region Montana Area Office

DRAFT

RESOURCE MANAGEMENT PLAN

Clark Canyon Reservoir and Barretts Diversion Dam

August 2004

Reclamation Office Contact:

Clark Canyon RMP Great Plains Region Montana Area Office 2900 4th Avenue North, Suite 501 P.O. Box 30137 Billings, MT 59107-0137

Prepared with Assistance from Franson Noble Engineering

Table of Contents

Cover Sheet Table of Contents Location Map Site Map

Chapter 1 – Introduction and Background

1.1	Executive Summary	. 1-1
1.2	Introduction	. 1-3
1.3	Policy and Resource Management Plan Framework	1-3
1.4	Need for Resource Management Plan	1-3
1.5	Purpose and Scope of the Plan	1-4
1.6	NEPA Compliance Requirements	1-4
1.7	Location and Description of the Management Area	1-5
1.8	Project History and Background	1-5

Chapter 2 – Existing Conditions

2.1 Physical Resource Conditions		2-1	
	2.1.1	Climate	2-1
	2.1.2	Geology	2-1
	2.1.3	Mineral Resources	2-3
	2.1.4	Soils	2-3
	2.1.5	Air Quality	2-4
	2.1.6	Water Quantity (Hydrology) and Water Quality	2-5
2.2	Biolog	gical Resource Conditions	2-10
	2.2.1	Vegetation	2-10
	2.2.2	Riparian Vegetation and Wetlands	2-10
	2.2.3	Noxious Weeds	2-11

Table of Contents (continued)

Page

	2.2.4	Wildlife	. 2-13
	2.2.5	Fish	. 2-14
	2.2.6	Threatened, Endangered and Sensitive Species	. 2-15
2.3	Aesthe	etic Resources (Visual and Noise)	. 2-16
	2.3.1	Noise Characteristics of the Project Area	. 2-16
	2.3.2	Visual Resources and Quality of the Project Area	. 2-18
2.4	Cultur	al Resources	. 2-21
	2.4.1	Introduction	. 2-21
	2.4.2	Prehistoric Period Resources	. 2-21
	2.4.3	Historic Period Resources	. 2-22
	2.4.4	Future Heritage Resources Focus	. 2-23
	2.4.5	Paleontological Resources	. 2-23
	2.4.6	Indian Trust Assets	. 2-23
	2.4.7	Indian Sacred Sites	. 2-23
2.5	Land U	Use and Management	. 2-24
	2.5.1	Clark Canyon Reservoir and Adjacent Lands	. 2-24
	2.5.2	Land Use Authorizations and Land Disposal	2-26
	2.5.3	Barretts Diversion Dam and Adjacent Lands	. 2-28
	2.5.4	Land Management Planning	. 2-28
	2.5.5	Other Resource Management Planning	. 2-28
	2.5.6	Montana Statewide Comprehensive Outdoor Recreation Plan (SCORP)	. 2-30
	2.5.7	Montana Fish, Wildlife and Parks	. 2-30
2.6	Recrea	ation and Visitation	. 2-30
	2.6.1	Fishing	. 2-31
	2.6.2	Boating	. 2-31
	2.6.3	Off-Highway Vehicle (ORV) Use	. 2-32
	2.6.4	Hiking	. 2-32
	2.6.5	Recreation Sites and Facilities	. 2-32

Table of Contents (continued)

Page

	2.6.6	Dispersed Sites	. 2-32
	2.6.7	Developed Sites	. 2-33
2.7	Transp	portation Resources	. 2-37
	2.7.1	Interstate 15 (I-15)	. 2-38
	2.7.2	State Route 324	. 2-38
	2.7.3	County Road 188 – Horse Prairie Cut-across Road	. 2-38
	2.7.4	Recreation Facility Access Roads	. 2-39
	2.7.5	Planned Improvements	. 2-39
	2.7.6	Access to East Side Lands	. 2-39
	2.7.7	Other Area Roads	. 2-40
	2.7.8	Airports	. 2-40
	2.7.9	Railroads	. 2-40
2.8	Socioe	economic Conditions	. 2-40
	2.8.1	Regional Setting	. 2-41
	2.8.2	Population	. 2-41
	2.8.3	Income	. 2-43
	2.8.4	Major Industries by Earnings	. 2-43
	2.8.5	Major Industries by Employment	. 2-44
	2.8.6	Unemployment	. 2-44
	2.8.7	Poverty	. 2-45
	2.8.8	Contribution of Recreation at Clark Canyon Reservoir to the Local Economy	. 2-45
	2.8.9	Net Benefits of Recreation	. 2-47
2.9	Public	Services and Utilities	. 2-47
	2.9.1	Beaverhead County Police Protection	. 2-47
	2.9.2	Emergency Medical Services	. 2-48
	2.9.3	Medical Facilities	. 2-48
	2.9.4	Volunteer Fire Districts and Companies	. 2-49

Page

2.9.5	Solid Waste	2-50
2.9.6	Electric Utilities	2-50

Chapter 3 – Needs, Constraints and Opportunities

3.1	Recreation Resources	3-1
3.2	Natural Resources	3-2
3.3	Lands	. 3-2
3.4	Cultural Resources	3-3
3.5	Operations, Maintenance, Enforcement, Public Health and Safety	3-4
3.6	Funding Opportunities	3-5

Chapter 4 – Resource Management Plan Actions and Elements

4.1	Plan Formulation		4-1
4.2	Recrea	tion Resources Goals and Objectives	4-3
	4.2.1	Barretts Diversion Dam	4-4
	4.2.2	Beaverhead River Fishing Access	4-4
	4.2.3	Buffalo Bridge Fishing Access	4-4
	4.2.4	Marina	4-4
	4.2.5	Lewis & Clark Campground	4-5
	4.2.6	Beaverhead Campground	4-5
	4.2.7	Camp Fortunate Overlook	4-5
	4.2.8	Sacajawea Day Use	4-5
	4.2.9	Horse Prairie Campground	4-6
	4.2.10	Cameahwait and West Cameahwait Campground	4-6

Table of Contents (continued)

Page

	4.2.11	Hap Hawkins Campground	4-6
	4.2.12	Lone Tree Campground	4-6
	4.2.13	Armstead Island	4-6
	4.2.14	South Shore Fishing Access	4-6
	4.2.15	Red Rock Fishing Access	4-6
4.3	Land l	Resources Goals and Objectives	4-7
	4.3.1	Horse Prairie Creek Mouth	4-8
	4.3.2	Red Rock River Mouth	4-8
	4.3.3	South Side Open Lands	4-8
	4.3.4	North Side Open Lands	4-9
	4.3.5	East Side Lands	4-9
	4.3.6	Beaverhead River Lands Below Dam	4-9
4.4	Cultur	al Resources Goals and Objectives	4-9
	4.4.1	Cultural Resource Goal	4-9
4.5	Opera	tions, Maintenance, Enforcement	4-9

Chapter 5 – Implementation and Monitoring Program

5.1	Alternatives Eliminated from Consideration	5-	-2
-----	--	----	----

Appendix A – Recreation Visitor Use Executive Summary

Chapter 6 – List of Preparers	6-1
Chapter 7 – References	7-1

List of Tables

Table No.	Table Title	Page
Table 2.1-1	Clark Canyon Reservoir Historical Monthly Flows	2-6
Table 2.1-2	Clark Canyon Reservoir Elevation Summary Table	2-6
Table 2.1-3	Clark Canyon Reservoir Monthly Discharge Flow Rate	2-7
Table 2.2-3	TES Wildlife Species Known or Expected to Occur	
	in the Vicinity of the Project	2-17
Table 2.2-4	TES Plant Species Known or Expected to Occur	
	in the Vicinity of the Project	2-19
Table 2.5-1	Land Ownership/Administration in Beaverhead County	2-27
Table 2.6-1	Clark Canyon Reservoir Recreation Facilities	2-37
Table 2.8-1	Estimated Driving Distances and Driving Time to	
	Clark Canyon Reservoir from Selected Cities	2-44
Table 2.8-2	Population of Selected Counties and Cities	2-45
Table 2.8-3	Earnings by Industry for 1999	2-46
Table 2.8-4	Employment by Industry for 1999	2-47
Table 2.8-5	Unemployment Rates	2-47
Table 2.8-6	Poverty Rates	2-48
Table 2.8-7	Poverty Levels	
Table 2.8-8	Recreation Contribution to the Local Economy	2-50

List of Figures

Figure No.	Figure Title	Page
Location Map		viii
Site Map		ix
Figure 2.1-1	Reservoir Elevations	
Figure 2.5-1	Land Ownership Map	
Figure 2.5-2	Barretts Diversion Dam Map	2-29
Figure 4.1-1	Clark Canyon Reservoir Proposed Plan	4-11





Clark Canyon Draft Resource Management Plan for Clark Canyon Reservoir and Barretts Diversion Dam

Chapter 1 Introduction and Background

1.0 Introduction and Background

1.1 Executive Summary

The Bureau of Reclamation (Reclamation) has prepared this Resource Management Plan (RMP) to provide prescribed guidance for management of natural and recreational resources, facilities, and access on Reclamation lands and waters surrounding Clark Canyon Reservoir and at Barrett's Diversion Dam in Beaverhead County, Montana. Clark Canyon Reservoir is situated alongside Interstate 15 approximately 18 miles south of Dillon, Montana. The RMP identifies goals and objectives for natural and recreation resource management, specifies desired land and resource patterns, and explains the policies and actions that will be implemented during the life of the plan to achieve those goals and objectives. **The RMP does not consider modifying reservoir operations**.

Clark Canyon Reservoir has a water surface area ranging from 4,500 acres at a typical low pool to over 5,900 acres at full pool that can be realized after wet winters with high spring runoff. The RMP is a land resource management plan focusing on how to best manage the 4,350-acres of Reclamation lands adjoining the reservoir and 38 acres of Reclamation land at Barrett's Diversion Dam, located 10 miles downstream of the reservoir. There are 11 developed recreation sites including 11 campground facilities around the reservoir and at Barretts Diversion Dam. These facilities attract approximately 57,000 visitors annually. Other than nearby foraging bald eagles, there are no other known occurrences of Federally listed Endangered or Threatened species at the project. Similar to most valleys in Beaverhead County there is however, a continuing problem with noxious weeds.

Development of the RMP began with a pre-planning process that included a recreation conditions assessment to establish current conditions for recreation use and facilities. That analysis culminated in a 2002 report; Recreation Management Condition Assessment. To solicit direct input from the public, agencies and other interested parties, Reclamation conducted a scoping process in fall of 2002 that included public meetings in Idaho Falls, Idaho, Dillon and Butte, Montana. Based on input gained from the scoping process along with the previous analyses, a series of alternative management and improvement measures were identified and evaluated in an Environmental Assessment that accompanies this draft RMP. The alternatives included No-action, representing a continuation of current management practices, moderate natural and recreation resource development that provides some improvements to most facilities around the reservoir, and a maximum resource development alternative that would bring all facilities up to current standards representing a close-to-build-out condition for the current facilities. As an outcome of the RMP analysis process and the Environmental Assessment, a single alternative, the Moderate natural and recreation Resource Development alternative was chosen to carry through the RMP based on several factors including demonstrated need, extent, type, and intensity of likely environmental and socioeconomic effects. During the development of the RMP, Reclamation hired the University of Mountain to conduct a visitor use survey. The survey obtained information on demographics, trip characteristics, and satisfaction about facilities, and whether there were conflicts or overcrowding. Data obtained in this survey were consistent with the data obtained at scoping meetings and alternatives studied in the RMP and EA. An Executive Summary of the results is included in Appendix A.

The RMP plan provides for a systematic level of site improvements at maintenance level quality across most existing developed facilities that are consistent with existing authorities demand, available funding, and carrying capacity of the resource base. Several sites would receive upgrades in the form of accessibility improvements that help bring facilities into compliance with the Americans with Disabilities Act and provide improved access to all types of users. Parking spaces and travel routes would be better managed at several sites. At some locations informal, user made roads would be closed (Sacajewea Day Use Area and Cameahwait campgrounds). At other locations parking lots would be delineated to provide more accommodations at peak times and lessen vegetative damage (Horse Prairie Campground and Buffalo Bridge Fishing Access). At Lone Tree Campground there would be improvements to the water supply facilities and new toilets would be added. New toilet facilities would also be added at Red Rock Fishing Access and Sacajawea Day Use Area. Environmental considerations include measures to reduce levels of ongoing erosion along portions of lower Horse Prairie Creek as well as adding additional jack-leg fencing at Horse Prairie Campground. Reclamation would continue to seek out a concessionaire to operate the marina facility while improving the facility by adding more shade trees and improving the water supply system.

For the open land resources outside of designated recreation areas, Reclamation would consider prescribed grazing for certain lands in the Horse Prairie Creek mouth to help revitalize the health of native vegetation and slow the ongoing spread of noxious weeds. Cottonwoods and willow trees would be planted to help maintain plant diversity in the riparian areas. Additional controls for noxious weeds would be taken along the abandoned railroad grade along in the Red Rock River area and fencing and signs would be added to other surrounding lands to exclude ORV uses. Sections of fence along the South Side Open lands would be replaced with more open fences friendly to antelope migration to improve access for this species.

Lastly, periodic monitoring will occur to provide periodic checks on the successes and outcomes of all management measures. The RMP plan is contemplated for a 10-year period and is designed to provide consistent planning direction to help coordinate land management activities with adjoining land owners and land management agencies.

1.2 Introduction

The Bureau of Reclamation (Reclamation) has prepared this Resource Management Plan (RMP) to provide prescribed guidance for management of resources, facilities, and access on Reclamation lands and waters surrounding Clark Canyon Reservoir and at Barretts Diversion Dam in Beaverhead County, Montana (Figure 1-1). This document is a draft RMP that is being circulated for public and agency review and comment. Following the receipt of public and agency input, Reclamation will finalize this document. Pending the completion of the associated environmental assessment and the signing of a Finding of No Significant Impact (FONSI), will adopt and implement this RMP.

Chapter 1 of the RMP provides the background and context for the plan. Chapter 2 presents detailed descriptions of existing resources and resource conditions at Clark Canyon Reservoir and Barretts Diversion Dam. Chapter 3 provides a summary of the constraints and opportunities associated with future uses, management and planning for the lands. Chapter 4 presents the proposed plan with a summary of its development and both specific and overall measures that provide planning direction for the lands over the 10-year planning period. Chapter 5 describes implementation measures and the types of monitoring that should be undertaken to determine if the plan and its measures are successfully, or are brining about the desired changes or outcomes. Lastly, the final sections of this RMP provide additional references and other materials to support future management decisions for the lands of Clark Canyon Reservoir and Barretts Diversion Dam.

1.3 Policies and Resource Management Plan Framework

Reclamation's mission statement is "to manage, develop, and protect water and related resources in an environmentally and economically sound manner in the interest of the American public" (Reclamation 2003). According to Reclamation's *2000-2005 Strategic Plan*, an RMP is to provide management direction consistent with authorized purposes while recognizing the rights and interests of existing contracts, legislation, and other entities concerning an identified land area that is under the jurisdiction of Reclamation. Recreation, Fish, and Wildlife are not authorized project purposes. They are indirect benefits of the project. Under P.L. 89-72, recreation is managed for minimum and basic facilities.

1.4 Need for Resource Management Plan



The RMP is intended to guide the future use, management, and site development of the lands around Clark Canyon Reservoir managed by Reclamation. Reclamation has the responsibility to manage its projects to meet its legal and contractual obligations, for the benefit of the public, and to protect natural, cultural, and recreational resources of those lands. An RMP identifies goals and objectives for resource management, specifies desired land and resource patterns, and explains the policies and actions that will be implemented during the life of the plan to achieve those goals and objectives. This RMP is a management tool for Reclamation, and other possible managing agencies, that outline resources, policies, and actions that will guide the agency over the 10-year life of the plan. Preparation of the RMP is specifically authorized in Title 28, Section 2805 (C) of Public Law 102-575.

In an August 2002, Reclamation conducted a Recreation Management Condition Assessment (Assessment Report) for Clark Canyon Reservoir and Barretts Diversion Dam. The Assessment Report found that these facilities are not utilized to their full recreation potential; therefore opportunities exist to provide additional or improved recreation opportunities and experiences to meet existing and future public recreation needs. The bicentennial celebration of the Lewis and Clark expedition, which is anticipated to bring thousands of new visitors to the area during the next five years, provide additional need for Reclamation to evaluate the adequacy of the current facilities, and implement an RMP that considers the likely increase in visitation.

1.5 Purpose and Scope of the Plan

The Clark Canyon Reservoir RMP is a 10-year plan to provide management direction for lands and recreational facilities under Reclamation jurisdiction at Clark Canyon Reservoir and Barretts Diversion Dam. The project includes land around Clark Canyon Reservoir and Barretts Diversion Dam, 10 river miles downstream of the reservoir. As previously stated, the purpose of the RMP is to address current and anticipated future issues to allow for orderly and coordinated development and management of lands and recreation facilities under Reclamation jurisdiction and authorities within the project area. Reclamation's broad goal for the RMP is to manage, protect, and enhance fish and wildlife habitat, natural resources, cultural resources, and recreation resources in a manner that is aesthetically pleasing and environmentally sound.

This RMP does not consider modifying reservoir levels or river outlet releases or water operations or address fisheries. Rather, the RMP is a land resource management plan focusing on how best to manage the lands and associated resources in a manner that is compatible and adaptable to the reservoir operations and its annually fluctuating water levels.

Through implementation of the RMP, Reclamation aims to balance competing and conflicting demands for differing uses of the land and maximize recreation and related activities compatible with surrounding land uses, while affording an appropriate level of resource protection and enhancement. Reclamation also intends to use this plan as a tool to seek out new managing partners for assistance in funding and operating recreation facilities and wildlife habitat areas.

1.6 NEPA Compliance Requirements

The National Environmental Policy Act (NEPA) of 1969 requires Reclamation to explore a range of possible alternative management approaches and the environmental effects of these actions. In conjunction with the preparation of this RMP, an Environmental Assessment (EA) is being prepared by Reclamation. The EA evaluates and compares three alternative management strategies: a moderate-level alternative (which is the proposed plan, as presented in Chapter 4 of this document), a high-level alternative, and a no action alternative. The impacts of each alternative are evaluated in the EA to identify potential impacts to resources including: water quality, soils, wetlands, vegetation,

biological resources, recreation, transportation and access, air quality, land use, noise, visual, cultural resources (including sacred sites and Indian Trust Assets), and socioeconomic conditions.

Where resource impacts are identified, the EA also recommends mitigation measures that serve to reduce or avoid potential adverse impacts. When appropriate, Reclamation has adopted such mitigation measures for inclusion in the RMP. Chapters 4 and 5 of this RMP discuss the RMP elements and identify mitigation measures that are also incorporated as elements of the RMP.

1.7 Locations and Description of the Management Area

Clark Canyon Reservoir is located in Beaverhead County in southwestern Montana. (See Location and Site map in front of document). The Clark Canyon Reservoir, which receives inflow from Red Rock River and Horse Prairie Creek, are the headwaters of the Beaverhead River. The project area includes Clark Canyon Reservoir and Barretts Diversion Dam, 11 river miles downstream of the reservoir. Clark Canyon Reservoir is situated alongside I-15 approximately 19 miles south of Dillon, Montana. Dillon, Montana is the county seat of Beaverhead County with a year 2000 population of 3,752 persons. The City of Idaho Falls, Idaho is 140 miles to the south and Butte, Montana is 85 miles to the north. Clark Canyon Reservoir has a storage capacity of 325,324 acrefeet of water with a normal pool capacity of 124,160 acre-feet that creates 5,903 acres of water surface area. The East Bench Irrigation District operates and maintains the dam, diversion facilities, and associated outlet works.

There are approximately 4,350 acres of Reclamation lands adjoining the reservoir surrounded by lands managed by the Bureau of Land Management (BLM), the State of Montana, and by private parties, with most of the surrounding lands used for ranching. The project lands around the reservoir, dam, and outlet structure facilities are currently managed for recreation, preservation of cultural resources, and wildlife habitat enhancement. The Barretts Diversion Dam, located just outside Dillon, Montana is situated on 38 acres of Reclamation lands and is managed for recreation use, cultural resources protection, and project-related uses.

1.8 Project History and Background

Clark Canyon Reservoir was developed in the 1960s as a part of the East Bench Unit of the Pick-Sloan Missouri Basin Program authorized by the Flood Control Act of 1944. The Project lands were originally acquired for the construction and maintenance of irrigation works and other purposes. Recreation and fish and wildlife were not authorized in the project legislation as an authorized purpose but are a secondary benefit of the project. The total area under Reclamation jurisdiction at Clark Canyon Reservoir consists of 5,903 water surface acres and 4,350 acres land surface acres immediately adjacent to Clark Canyon Reservoir. (These averages are for when the reservoir is at a "normal pool" elevation of 5536 feet above mean sea level [msl]) In addition, there are approximately 38 acres available for public use at Barretts Diversion Dam. This Recreation Management Condition Assessment dealt primarily with assessing the current recreation conditions. It identified existing issues and concerns related to recreation management. Possible development alternatives, constraints and opportunities were identified. This Report served as the foundation for formulating many of the RMP and Environmental Assessment (EA) recommendations and reference material.

Prior to 1978 the Montana Fish, Wildlife and Parks (MFWP) managed the recreation and wildlife resources at Clark Canyon Reservoir but since that time Reclamation has managed these resources through its limited authority under Public Law 89-72. Under Public Law 89-72, Reclamation is authorized to provide only minimum basic facilities (which at Clark Canyon Reservoir consists of general maintenance of existing facilities).

Construction of Clark Canyon Dam was started in 1961 and completed in 1964 as an earth filled structure. Clark Canyon Dam was authorized as part of the Pick-Sloan Missouri Basin Program, East Bench Unit. Horse Prairie Creek and Red Rock River provide the water used for storage in the reservoir. There are three direct benefits associated with Clark Canyon Dam. The first and most important direct benefit is: water supply for increased irrigated land, which results in an increase of net farm income and in agricultural production. Operation of the reservoir is controlled by a contract to deliver the necessary water to the irrigated land. Second, Clark Canyon reservoir provides extensive flood control on the Beaverhead River. Third, the reservoir provides extensive recreational opportunities, even though recreation is not an authorized project purpose. The recreational opportunities are dependant on the water level of the reservoir, which fluctuates depending on precipitation. Since recreation was not authorized, there is no storage reserved to provide recreational opportunities. The capacity of the reservoir is 257,152 acre-ft. The surface area of the reservoir at capacity covers 5,903 acres. The length of the crest is 2,050 ft. The height of the dam is 147.5 ft.

Another feature of the East Bench Unit is Barretts Diversion Dam. The diversion is used to feed the East Bench Canal and the Canyon Canal. These canals provide water for 28,000 acres of irrigated farmland in Madison and Beaverhead Counties. Excess water from the canals and return flows return to the Beaverhead River.

After construction of the dam, operation and maintenance of the recreational facilities surrounding the reservoir was conducted by MFWP through a management agreement with Reclamation. Improvements were made under the direction of MFWP from 1964 until 1978. In 1978, MFWP returned management of the recreation areas back to Reclamation. Since then, existing facilities have been operated and maintained by Reclamation through limited authority under Public Law 89-72. Other than replacement of antiquated facilities, no new recreation facilities have been constructed since mid-1970.



One important aspect of the Clark Canyon Reservoir and Barretts Diversion location are that they are closely associated with the Lewis and Clark Expedition of 1804-1806. Camp Fortunate, presently inundated beneath the reservoir, is where Captain Lewis first made contact with the Shoshone Indians. Camp Fortunate is also where the Shoshone first recognized Sacajawea, and where she was united with her brother, Cameahwait, the leader of the band. Sacajawea's relationship to the Shoshone and the Lewis and Clark expedition was important to the success

of the expedition. The Shoshone offered food and horses, which the expedition critically needed in order to move further west in their quest for a route to the Pacific Ocean.

In addition, Barretts Diversion Dam is also a recognized landmark documented in the Lewis and Clark diaries as Rattlesnake Cliffs by the number of rattlesnakes they encountered. These aspects of the local history have national importance, as there is likely to be an increase in tourism and visitor use in the area as a result of the increased interest in Lewis and Clark bicentennial celebration.

Clark Canyon Draft Resource Management Plan for Clark Canyon Reservoir and Barretts Diversion Dam

Chapter 2 Existing Conditions

2.0 Existing Conditions

This Chapter discusses the existing physical, biological, economic, and other characteristics of the project area and region. The consideration of these existing conditions provide the context within which Reclamation has developed and will implement the RMP, and are therefore important to present here to provide a general understanding of the general conditions of the project area.

2.1 Physical Resource Conditions

2.1.1 Climate

The climate of the South Central Mountains can be broadly described as continental and is typical of the high elevation arid lands in the region, with wide variability between summer and winter. Occasionally, cold arid arctic air moves down over the area creating winter temperatures below zero.

The average annual precipitation measured in Dillon, Montana has been approximately 11.6 inches in the last thirty years (1971 - 2000). There has been a significant drought for the past 5 years (1999-2003). Between 1900 and 1925, the average rainfall was 18.3 with every year experiencing more rainfall than the recent average of 11.6 inches. For a longer historical period (1901-1940), the average annual precipitation at Dillon was 16.67 inches. The wettest months are generally May or June with approximately 75 percent of the precipitation falling in the six months, April to September. There is a wide range in the annual rainfall with 6.9 inches in 1974 and 19.2 inches in 1995 (WRCC 2002).

The average annual temperature varies widely by season and within each month. The hottest months are July and August with average maximum temperature of 84° and 82° F and average minimums of 47° and 46° F, respectively. The coldest months are normally December and January with average maximum temperatures of 34° F and average minimum of 14° F (WRCC 2002).

The closest national weather station is located in Dillon, Montana, which is 19 miles north of Clark Canyon Reservoir. The temperatures at the reservoir tend to be slightly cooler than those reported near Dillon; however the area around the reservoir experiences fewer frost-free days than Dillon. There are approximately 70 to 90 frost free days at the reservoir compared to 103 frost free days in Dillon (USDA 2002, and personal conversation, Gary Berger, Natural Resources Conservation Service, Dillon, MT, 10/4/2002). Frequently, the reservoir also experiences high winds limiting boat launching and other outdoor activities.

2.1.2 Geology

The geologic history of southwestern Montana is complex and is of regional interest for geologists. The geology was shaped by processes following the intense mountain building activity that resulted in the Rocky Mountains. Regional compressions and extensions resulted in a system of northeast and northwest trending faults and basins such as the Beaverhead Basin. Great volumes of sediment washed into basins from surrounding mountains. Volcanic activity blanketed the area with ash and produced fields of volcanic rocks, some with substantial quantities of andesite. Climatic changes also produced geologic affects. Tropical periods with lush vegetation prevented widespread erosion

and the streams flowed in more permanent channels and carried fine-grained sediments. During arid periods permanent streams became ephemeral and there was less vegetation to maintain the channel and to prevent erosion. As a result coarser-grained sediments were deposited in flash floods that spread sediments across the landscape. While the landscape continues to be modified, it is much less dramatic. The streams create alluvial deposits, and landslides have occurred along steep slopes of clayey materials. (Garsjo 2002, Alt and Hyndman 1986).



At Clark Canyon Reservoir the geology is also complex. Rocks in the area show a very wide variety of ages, from pre-Cambrian to Quaternary eras. Northwest trending thrust faults intersect the older sediments that are folded into a series of north trending synclines and anticlines. The Tendoy thrust fault separates the south plunging Armstead anticline in the Red Rock basin from the Clark Canyon Syncline. Gentle folding, high-angle block faults influenced the drainage system and there were three major erosional cycles creating thick sediments, terraces, and modern stream flats.

The result is that the east boundary of the reservoir is gently sloping Tertiary conglomerate sandstone and Quaternary gravels. The northwest portion of Clark Canyon Reservoir has formations ranging from Pre-Cambrian gneiss and Paleozoic rocks, limestone, granite, and quartzite. The southwest boundary is composed of Quaternary sediments. These are mapped as primarily Quaternary terrace gravels and alluvial fans, but include limestone and quartzite formations. The drill logs of the area show gravels in thicknesses of up to more than 175 feet in areas at the reservoir site.

There are visible geologic resources of interest to people driving on I-15 from Monida Pass north through Montana. In the immediate area of the reservoir are two notable examples. East of the reservoir, across the highway, is an expanse of Beaverhead conglomerate, a formation that is very limited in the region of southwest Montana.

The second example is found at the Tendoy Range, west of the reservoir. The edge of the Tendoy Range is defined by the Red Rock Fault. The fault creates a straight face to the range that is visible from the reservoir (Alt and Hyndman 1986).

Further north, close to Barretts Diversion Dam, are volcanic rocks (white rhyolite ash and red brown andesite) that are thought to be from two generations of volcanism, 50 and 70 million years ago. The reddish brown hill, located about 10 miles north of Dillon, is possibly the one that Clark identified in his journal as Beaverhead Rock (Alt and Hyndman 1986). It is volcanic andesite and could have been the landmark that helped Sacajawea guide the expedition. The Rattlesnake Cliffs, north of Barretts Diversion Dam were named by Lewis in 1805 and are composed of andesite up thrown on the Blacktail fault (Garsjo 2002). The cliffs form the entrance to the Beaverhead River Canyon.

Montana has a history of large and damaging earthquakes. Several active earthquake zones are located in southwest Montana. The nearest fault zone to Clark Canyon Reservoir is the Red Rock valley, a NW-trending graben, which extends from Lima to the Clark Canyon Reservoir (Stickney, M.C. 1995). A graben is a down-dropped block of the earth's crust resulting from extension, or pulling, of the crust. The east-dipping Red Rock fault bounds the southwest valley margin but extends only half the distance northwest to Clark Canyon Reservoir. The central portion fault zones are the most seismically active segment of the Red Rock valley region and have experienced earthquakes clusters in 1985 and 1996 along with frequent micro earthquake activity. The 1985 clusters occurred prior to good regional seismic instrumentation, and few events were suitably recorded for first motion analysis.

Two earthquakes occurred at opposite ends of the Red Rock valley graben in April 1991 and March 1994. Despite the lack of a detailed mapping of the graben separation, the March 1994 earthquake indicated that the northern part of the Red Rock graben experienced NE-SW faulting

2.1.3 Mineral Resources

Many energy and mineral resources occur on both private and government-managed lands within Beaverhead County. These resources have great economic potential for the local area (BCRUC 2003). However, there are no known mineral deposits located in the project area except for sand and gravel. Historically, there had been sand and gravel mining in the reservoir area for local use (Holt 1956). Sand and gravel deposits will be used by Reclamation on an as needed basis for project related purposes; such as construction and maintenance of dams, facilities, and recreation areas. The Bureau of Land Management manages all subsurface mineral resources (excluding sand and gravel) on the land owned by Reclamation around Clark Canyon Reservoir.

2.1.4 Soils

The area surrounding Clark Canyon Reservoir is generally characterized by a series of intermountain valleys characteristic of the topography of southwestern Montana. The Blacktail Range bound the valley to the east and northeast, by the Pioneer range to the northwest, and the Tendoy Range to the southwest.

The soils at Barretts Diversion Dam, at the mouth of the Beaverhead River Canyon, are in the process of being mapped by the National Resources Conservation Service (NRCS). The soils are dominated by Rivra, Cool-Beavrock Complex (USDA NRCS 2002). The complex is approximately 45 percent Rivra and similar soils, and 40 percent Beavrock and similar soils. Rivra soils are characterized as gravelly sandy loam and well drained with frequent flooding. The Beavrock soils are silt loam and poorly drained with occasional flooding.



Clark Canyon Reservoir has not been mapped by NRCS but can be generally described from a few sources. The 1956 assessment of fish and wildlife for the construction of the dam describes the soils

as brown or grayish-brown, granular, gravelly loam over gravelly subsoil (USBR, 1956). The valleys are filled with lacustrine and fluviatile deposits. The mountain slopes in the area around Clark Canyon Reservoir are mostly covered by alluvial fans and alluvial deposits derived from the underlying rocks. Limited exposure of bedrock found in cliffs and canyons reveal igneous sedimentary and metamorphic rocks.

Core sampling done in 2001 to investigate the Clark Canyon landslide located approximately 4 miles north of the dam described the upper feet of the core sample to be generally silty, sandy, and gravelly (McDonald 2001).

2.1.5 Air Quality

According to the U.S. Environmental Protection Agency's (USEPA) air database and the State of Montana's Department of Environmental Quality (MDEQ), there are 13 areas in the state of Montana that are classified as "non-attainment". Non-attainment refers to areas of the country where air pollution levels persistently exceed the national ambient air quality standards. Particulate matter is the most common cause of non-attainment in Montana; however, Billings is considered a non-attainment area for sulfur dioxide (MDEQ, 2003). The non-attainment areas are centered primarily on high population areas, including the Missoula area in the northwest portion of the state. USEPAS Monitor Trends Report (2003) indicates that the number of days when air quality within monitored counties throughout the state was "good" or "moderate" far exceeded the number of days when air quality was classified as "unhealthy".

Air quality at various locations within the state of Montana is monitored and data is collected by the MDEQ. According to the MDEQ, the principal air quality concern in Beaverhead County is PM_{10} (particulate matter less than ten microns in diameter), which may have harmful effects on the human respiratory system. MDEQ also has noted that many federal, tribal, and state forestry managers increase the use of prescribed burns to create resource benefits on public lands. Although this practice can benefit habitat, the natural by-product of the burning is smoke, which is primarily made up of particulate matter (MDEQ, 2003).

The primary contributor to PM_{10} in the Dillon area appears to be associated with industrial activity six miles south of Dillon around the Barretts Minerals. Inc., a talc plant (which is approximately 0.5 miles north of Barretts Diversion Dam). In September 1984, a monitoring site was established at the Beaverhead County Courthouse (Monitoring Site #30-001-0001). This site, the nearest air quality monitoring station to the project area, operated for two years until June 1986. The monitoring data showed relatively low readings and it was determined that continued monitoring was not warranted. As such, only limited air quality data specific to the project area is available.

The most likely contributors to air pollutants within the project area are motor vehicle emissions (including automobile and truck traffic on I-15 and other area roadways, as well as, boat and personal watercraft (PWC) operation on Clark Canyon Reservoir), diesel-engine trains that use the railway, which passes east of Clark Canyon Reservoir and west of Barretts Diversion Dam, fugitive dust emissions (primarily from motor vehicle operation on unpaved roads), smoke from wood burning stoves, fireplaces, and campground fires, and controlled burning of open land.

<u>Air Quality Regulation</u> - Under the Clean Air Action of 1970, the U.S. EPA developed primary and secondary National Ambient Air Quality Standards (NAAQS) for each of the seven criteria

pollutants: carbon monoxide, lead, nitrogen dioxide, ozone, particulate matter, fine particulate matter, and sulfur dioxide. These standards establish pollution levels in the United States that cannot legally be exceeded during a specified time period. The State of Montana has adopted additional state air quality standards. These Montana Ambient Air Quality Standards (MAAQS) establish statewide targets for acceptable amount of ambient air pollutants to protect human health. In addition, congress also classified select regions of the country as areas designed as Prevention of Significant Deterioration (PSD). The classification differs in the amount of development allowable within the area. In Montana, the only areas designated as Class I areas (the most restrictive limits on development) are two National Parks, several National Wilderness areas, and several Native American Reservations. The nearest Class I area is Red Rocks Lakes, located approximately 100 miles southeast of Clark Canyon Reservoir near the southeast border of Beaverhead County and Madison County.

2.1.6 Water Quantity (Hydrology) and Water Quality

<u>Water Quantity</u> - Clark Canyon Reservoir was built in 1964 primarily for irrigation purposes. The East Bench near Dillon needed a reliable source of irrigation water, and the reservoir was designed to provide full irrigation water to 21,800 acres and supplemental irrigation water for an additional 28,000 acres. Secondary purposes of the reservoir include flood control and recreation.

Clark Canyon Dam has a total drainage area of 2,315 square miles and the two main sources of inflow to the reservoir are Red Rock River and Horse Prairie Creek (as shown on Location Map). During the 39 years of record since the construction of the Dam, the average annual inflow into the reservoir is 274,500 acre-feet (see Table 2.1-1). In 1984, the maximum yearly inflow occurred with a total of approximately 718,100 acre-feet. The minimum yearly inflow occurred in 2003 with a total of approximately 105,189 acre-feet.

The elevation of the reservoir is controlled by the operating plan established for Clark Canyon Reservoir. There is some target elevations used to manage the reservoir and to allow for protection against naturally occurring floods. The reservoir management plan calls for the storage between elevations 5542.1 above mean sea level (msl) and the spillway elevation of 5560.4 msl to be exclusively used for flood control. Storage between elevations 5535.7 and 5542.1 msl is referred to as the joint use pool and can be used for irrigation storage as well as flood control. The surface area of the reservoir at the top of the active conservation pool is 4,496 acres at elevation of 5535.7 msl. Clark Canyon Reservoir has reached this elevation in 32 of the 39 years of operation. At the top of the joint use pool, the surface area of the reservoir is 4,935 acres at elevation of 5542.1 msl. The reservoir has reached this elevation of 5560.4 msl. The reservoir has reached this elevati



Table 2.1-2	
Clark Canyon Reservoir Elevation Summary	Table

			Storage	Surface
Description	Elevation	Probability	Capacity	Area
	(ft)		(acre-feet)	(acres)
Maximum Water Surface	5571.9	0%	328,979	6,600
Spillway Crest	5560.4	3%	257,152	5,903
Top of Joint Use Pool	5542.1	79%	157,863	4,935
Top of Active Conservation Pool	5535.7	82%	127,626	4,496
Top of Inactive Storage Pool	5470.6	100%	1,500	209

When irrigation season begins approximately (May 1^{st}), the average level of the reservoir is 5,540.9 msl. The surface area of the reservoir at this elevation is 4,868 acres. In a normal (or average) year, the reservoir drops approximately 10 feet during the irrigation season. At the end of irrigation season (September 30^{th}), the average level of the reservoir is 5,530.9 msl. The surface area of the reservoir at this elevation is 4,109 acres. During most years, approximately 759 acres of shoreline are exposed during the irrigation season. See Figure 2.1-1 to see the surface area of the reservoir at elevations discussed above.

As of early April 2004, the approximate elevation of the reservoir is approximately at elevation 5,516 msl. The western United States is in the 5th year of a drought cycle. The drought has taken its toll on the storage and consequently the elevation of the reservoir. Due to the drought, the reservoir is at its lowest level since Clark Canyon Reservoir was filled in 1964. The surface area of the reservoir at elevation 5,490 msl is 937 acres. This change in elevation of the reservoir had the greatest impact on the south and southwest shorelines. Because of the gradual slopes in these areas, a slight change in reservoir elevation translates into longer stretches of shoreline that are exposed. The least impact of this drought cycle has occurred on the northwest shoreline. The slope into the reservoir is more pronounced as a change in water level has a reduced effect on exposed shoreline.

The minimum releases from the reservoir have been adjusted this year (2003) to retain as much irrigation water as possible in the reservoir. In August 2003, the minimum flow released from the reservoir was 25 cubic feet per second (cfs). Table 2.1-3 shows monthly average, maximum and minimum discharge flow rates for the reservoir.



The average stream flow below the Clark Canyon Reservoir at the beginning of the irrigation season on approximately May 1 is 500 cfs. It normally peaks in July at 775 cfs and reduces to 475 at the end of September. During the non-irrigation season of October 1 thru April 30, the average stream flow is 475 cfs to 200 cfs with a minimum of 25 cfs. During the last couple of drought years, the minimum flow has been 25 cfs.





Since the construction of the Clark Canyon Reservoir flooding has not been a problem and high flows have stayed within the banks of the Beaverhead River downstream of the dam.

<u>Water Quality</u> - Water quality within Clark Canyon Reservoir is affected by several factors, including water quality in the two primary tributaries, Red Rock River and Horse Prairie Creek; water levels within the reservoir; and contributing factors from land uses surrounding the lake. The water quality within Clark Canyon Reservoir is safe for all contact sports and recreation and is considered to be "clean", as compared to other similar reservoirs.

Water quality sampling at Clark Canyon Reservoir began in 2001 and continued through 2003. The general purpose of the sampling has been to identify potential problems and to establish baseline data, which will be compared to future data to determine changes in the reservoir. Five sites were sampled within the reservoir, as well as both inflows (Red Rock River and Horse Prairie Creek and the Beaverhead river immediately below the dam). Sampling occurred on a monthly basis beginning in April and continuing through October.

Basic limnological evaluations were completed at each site within the reservoir, consisting of water column profiles, depth, nutrients, chlorophyll, phytoplankton, and zooplankton samples. Water column profiles recorded water quality parameters for dissolved oxygen, specific conductance, pH and temperature from surface to bottom. Zero to 5 meter integrated samples was collected for chlorophyll analysis. Integrated samples of phytoplankton and 0-15m integrated samples of zooplankton were collected at 0-5m at each site for determining the number of species and density. Nutrient grab samples were collected from both the hypolimnion and epilimnion (i.e. near the bottom and top) of the lake. Samples were analyzed for ortho-phosphate, nitrate, ammonia, and nitrogen. Sediment samples for trace metals were collected in 2003. Samples were analyzed for identification and quantification of total extractable metals. The findings of the study should be summarized in a report by September 2004.

The inflows and outflow of Clark Canyon Reservoir water quality parameters tested for included temperature, dissolved oxygen, specific conductance, pH, ortho-phosphate, nitrate, ammonia, and nitrogen. All samples were taken in shallow (<1m) moving water near the shoreline.

The Clean Water Act requires States to set water quality standards and enforce them. Flowing out of Clark Canyon Reservoir, the Beaverhead River is listed on the State of Montana's 2002 303(d) List of Impaired Water Bodies for the segment from Clark Canyon Dam to Grasshopper Creek. This segment does not meet Montana State Water Quality Standards. The factors contributing to this listing include bank erosion, dewatering, flow alteration, and inflows from agricultural and mining sources. In addition, Horse Prairie Creek from the headwater to Clark Canyon Reservoir and a portion of Red Rock River from Lima Reservoir to Clark Canyon Dam are also included on the state of Montana Department of Environmental Quality 303 (d) list of impaired water bodies.

2.2 Biological Resource Conditions

2.2.1 Vegetation

The Clark Canyon Reservoir is east of the Continental Divide with a continental climate. It is typical of the high, arid lands in the region, with wide variability between summer and winter. The Reservoir area is in the Beaverhead Mountains Section (Section M332E) of the Northern Rocky Mountain physiographic province (McNab 1996). The Beaverhead Mountains Section is generally mountainous with broad valley bottoms. Vegetation in the section consists of two primary types: 1) sagebrush steppe with small areas of alpine vegetation above 9,500 feet (75 percent) and 2) Douglas-fir forest (25 percent).

Clark Canyon Reservoir is in a valley bounded by the Blacktail Range to the east and northeast, by the Pioneer range to the northwest, and the Tendoy Range to the southwest. The reservoir has approximately 5,903 acres of surface area and 17 miles of shoreline at 5535 feet (top of active conservation pool elevation). The project area covers approximately 4,350 land acres surrounding the reservoir and owned by the Reclamation (USDOI 2002). Of the land acres, approximately 150 (less than five percent) are developed for campgrounds, picnic areas, and boat launches. Much of the area was once irrigated pastured or hayed before the reservoir existed.

The land around the reservoir is one of the westernmost examples of short grass prairie habitat. Short grass prairie habitat is more common further east of the mountains. The grasses are primarily native bunch grasses and wheat grasses. There are swales in the reservoir land with sagebrush communities but extensive sagebrush steppe is more prevalent further toward the mountains. Mud flats appear in the reservoir seasonally as the water recedes. They generally develop populations of leafy spurge, a noxious weed.

2.2.2 Riparian Vegetation and Wetlands

As previously stated, the two main watercourses that supply water to the Clark Canyon Reservoir are the Red Rock River to the south and the Horse Prairie Creek to the west. The reservoir drains to Beaverhead River. Willows, cattail marshes, and wet and semi-wet meadows border these three watercourses and riparian areas. Riparian habitat and cattail marsh wetlands only cover small portions of the land in southwest Montana (less than one percent) but support much of the wildlife.



The Red Rock River is south of Clark Canyon Reservoir and is located in the wide valley bottom. The river meanders through an emergent wetland and varies in size depending on runoff. The Red Rock River riparian area is predominately willow with adjacent semi-wet meadow including rushes and sedges. Based on field observations, there appears to be a buildup of plant residue in the riparian area and there are large patches of the noxious weed Canada thistle near the south boundary of the Reclamation land. West of the Red Rock Creek riparian zone, is a monoculture growth of reed canary grass. The reed canary grass appears to cover the area from the mid-reservoir water level to the high water level. The reed canary grass also appears to include a buildup of plant residue. Much of the area adjacent to Horse Prairie Creek was once hay land or irrigated pasture, however, it has not been irrigated or grazed by domestic livestock since Clark Canyon reservoir was built in 1964. Based on field observations, there also appears to be a buildup of plant residue and there are patches of the noxious weeds, Canada thistle, and whitetop.

Mud flats are located along segments of the reservoir during periods of low water. These mud flats provide habitat for shore birds such as marbled godwits, long-billed dowitchers, willets, western sandpiper, and spotted sandpiper. Pelicans have been seen in late summer (personal conversation J. Kirkley, University of Montana, Dillon, 4/17/03).

Barrett's Diversion Dam has combined willow and cottonwood riparian habitat and also has a cattail marsh. The tree, shrub, marsh combination, together with rock faces on both sides of the adjacent road, provide a variety of habitats for bird nesting and foraging.

2.2.3 Noxious Weeds

Noxious weeds can be a serious environmental problem to natural resources. Noxious weeds can displace native plant communities (including endangered species), alter wildlife habitat, reduce forage for wildlife and livestock, increase erosion, and lower biodiversity. The Federal Noxious Weed Act of 1974 requires that each federal agency develop a management program with adequate funding to control undesirable plants on lands under its jurisdiction. It also requires that the agencies implement cooperative agreements with state agencies to coordinate management of undesirable plants.

Executive Order 13112 of 1999 was enacted to "prevent the introduction of invasive species and provide for their control and to minimize the economic, ecological, and human health impacts". In this act, federal agencies were directed to: 1) prevent introduction of invasive species, 2) detect and respond to control such species in a cost-effective and environmentally sound manner, 3) monitor invasive species populations accurately and reliably, 4) provide for restoration of native species and habitats, 5) conduct research on invasive species. Furthermore, they were directed to not authorize, fund or carry out actions that are likely to cause or promote invasive species. Federal agencies are also required to pursue these duties in cooperation with stakeholders.

The Montana County Noxious Weed Control Law (MCA 7-2101 through 2153) was established in 1948 and amended in 1991, to protect Montana from destructive noxious weeds. It established a set of criteria for the control and management of noxious weeds in Montana and is implemented and enforced at the local county level through county weed control boards. Montana law makes it unlawful for any landowner to permit noxious weeds to propagate or produce seeds on their land unless they adhere to the noxious weed management program of their county. While Montana laws do not apply to federal lands, the Federal Noxious Weed Act requires cooperative planning with the state/local weed administration.

Beaverhead County Weed District is active in highway weed management and works with the Reclamation and other landowners to control noxious weeds. Table 2.2-1 is noxious weeds listed by the state and Beaverhead County Weed District. The noxious weeds that are targeted for containment and suppression on the Clark Canyon Reservoir resource area are: whitetop (hoary cress), Canadian thistle (*Cirsium arvense*), leafy spurge (*Euphorbia esula*), and spotted knapweed (*Centaurea maculosa*) (personal communications Jack Eddie, Beaverhead County Weed District,
Dillon MT) and are described below. Descriptions for these species are described below. They are all Category One Noxious Weeds, which are defined by Montana's Weeds Control Law as: Currently established and generally widespread in many counties of the state.

Whitetop (Hoary cress) (Cardaria draba)

Whitetop is a deep-rooted perennial plant that reproduces from root segments and seeds. Plants emerge in early spring and bloom and set seed by midsummer. This plant is common on alkaline, disturbed soils and is highly competitive with other species once it becomes established. Control is difficult because of the perennial root system, abundant seed production, and diverse habitat of the plant. (Esser 1994)

Whitetop may have some forage value but the foliage becomes coarse and bitter as it matures. Until research is able to better evaluate its toxicity, McInnis (<u>McInnis</u>



et. al. 1993) recommends the following cautionary measures when allowing livestock to graze infested rangelands: provide supplemental iodine, graze mature and non-lactating animals, and provide alternate grazing areas.

Canada thistle (Cirsium arvense)

Canada thistle is a non-native, colony-forming perennial with deep and extensive horizontal roots. Canada thistle reproduces both through seed and asexually with spreading roots, root fragments, or from subterranean stem tissue. It has the potential to rapidly infest an area through vegetative reproduction. This aggressive weed is difficult to control. Flowering occurs during July and August. Canada thistle is generally not considered palatable by livestock. They generally avoid Canada thistle and may also avoid grazing in areas with heavy infestations. Fire may kill the above ground portion of Canada thistle, but the below ground portion can survive even severe fires (Zouhar 2001a).

Leafy spurge (*Euphorbia esula*)

Leafy spurge is a perennial that reproduces by vigorous rhizomes and seed. It generally grows in clumps, up to 3 feet tall. Leafy spurge has an extensive underground rhizome and root system that can penetrate the soil to 15 feet to 30 feet deep. The deep rooting and thick, corky root bark help give the species drought resistance. The plants store carbohydrates and nitrogen within the rhizomes during fall and winter. Reserves accumulate in the fall as top-growth dies and are in low supply during the growing season. The entire plant contains a milky juice that has been reported to cause severe irritation of the mouth and digestive tract in cattle and may result in death. Cattle prefer to graze where there are not large infestations. The extensive root system containing large nutrient reserves makes leafy spurge extremely difficult to control (Simonin 2000).

Spotted knapweed (Centaurea marculosa)

Spotted knapweed is a biennial or perennial with a stout taproot. Spotted knapweed reproduces almost entirely from seeds that is known for their longevity and durability. Knapweeds readily establish themselves on any disturbed soil, and their early spring growth makes them competitive for soil moisture and nutrients. Spotted knapweed has good nutritional value and is eaten by livestock

and wildlife. Generally, use of spotted knapweed is highest during early season when the plants are green and actively growing in the rosette and bolt stages. However, it can impede access to other forage when there are old stems that form a spiny over story (Zouhar 2001b).

Clark Canyon management priorities currently include awareness and education, containment and suppression of existing infestations, and the prevention of new infestations. Beaverhead County is contracted by the Reclamation to treat the noxious weeds at Clark Canyon Reservoir and Barrett's Diversion Dam. The County has also posted educational signs in several places around the reservoir. The noxious weed infestations that receive treatments are around the campgrounds and Beaverhead River below the dam. There is also a large infestation of leafy spurge in the area of the reservoir that is exposed due to low water (personal communication J. Eddie, Beaverhead County Weed District, Dillon, MT, 4/15/03).

While bio-control has been used in the past to treat leafy spurge, current weed treatment consists of herbicide application. In 2000 and 2001, weed treatment was reduced by at least half resulting in a notable spread in weeds (personal communication S. Morehouse, Bureau of Reclamation, Dillon, MT, 4/31/03). Beaverhead County Weed District resumed full treatment in 2002 and has increased its activities by obtaining a grant from the Montana Noxious Weed Trust Fund. The grant provides for up to five years of additional treatment on spotted knapweed that is 300 feet each side of the Beaverhead River on Reclamation property below the dam (personal communication J. Eddie, Beaverhead County Weed District, Dillon, MT, 4/15/03). The determination of effectiveness of noxious weed treatments is based on observation. The current program has not been monitored and there is not a clear determination of whether the noxious weeds are being contained or not.

2.2.4 Wildlife

As described in section 2.2.1, Vegetation, Clark Canyon Reservoir lands are primarily short grass prairie with areas of sagebrush habitat in the uplands. There are three riparian areas with willow and cottonwood habitat, cattail marshes, wet and semi-wet meadows. As stated earlier, mud flats are exposed during period of low water. The flats provide habitat for shore birds such as marbled godwits, long-billed dowitchers, willets, western sandpiper, and spotted sandpiper. Pelicans have been seen in late summer (personal conversation J. Kirkley, University of Montana, Dillon, 4/17/03). These diverse habitats around Clark Canyon support many bird and animal species such as golden eagles, ferruginous hawk, chestnut-collared longspurs, antelope, and white tailed deer. The sagebrush in the drainages also provides habitat for pygmy rabbits.



The wet and semi-wet meadows and cattail and willow wetland adjacent to the streams provide forage and nesting habitat for several migrant birds including yellow-headed blackbird, red-winged blackbird, common snipe, Wilson's phalarope, rails, sora, and several species of ducks. The cattail marsh, which is located on the Beaverhead River immediately below the dam, is spring fed and has a

constant temperature of 54° Fahrenheit (12° C). It remains unfrozen and provides year round refuge for a myriad of wildlife species.

The portion of Beaverhead River immediately below the dam is diverse and well suited for viewing wildlife, springs, river, and plant diversity. It has been identified as a Montana Wildlife Viewing Area for the Montana State Watchable Wildlife Program by a panel of wildlife experts from state and federal organizations. The goals of the program are to: 1) Identify and enhance opportunities for the public to enjoy wildlife on public and private lands; 2) Contribute to local economic development; 3) Promote learning about wildlife and habitat needs; and 4) Enhance active public support for resource conservation selected all sites. There is signage on the highway identifying the site and interpretive signage at the site. It is also listed in tour books as a Montana Wildlife Viewing Area.

The reservoir is in the Beaverhead/Red Rock Corridor flyway for migrating birds. It is a known resting or foraging spot for many migrant birds. During spring migration, loons are seen in the reservoir, courting behavior has been observed, and a nesting pair has been documented (personal conversation S. Morehouse, Bureau of Reclamation, Dillon, MT). Snow geese, trumpeter swans, and tundra swans often stay in the reservoir area for days or weeks, depending on surface water area, as they migrate through the area to nest in Red Rock Lakes National Wildlife Refuge to the southwest.

Horse Prairie Creek is in a narrower valley than Red Rock River and the creek and has a more direct course. Montana State Route 324 (SR 324) runs west from the reservoir along Horse Prairie Creek. There is also a remnant of an old railroad grade in the drainage. The area is west of the Clark Canyon Reservoir and is generally managed for wildlife habitat. Based on field observation and personal conversation with B. Gibbons (Rangeland Management Specialist, NRCS Dillon Montana, 6/19/03) it appears that the willows and cottonwood north of SR 324 are in poor condition with many already dead. This appears to be a result of channel changes and the recent drought that have left the plants on dry ground. The new channel does not have new cottonwoods and has only a few shrubs and willows components in the riparian vegetation.

At Barrett's Diversion Dam, the tree, shrub, marsh combination, together with rock faces on both sides of the adjacent road, provide a variety of habitats for nesting and foraging.

2.2.5 Fish

The Clark Canyon Reservoir provides a diversity of fish habitat with the lake environment and three streams. The Montana Fish and Wildlife and Parks (MTFWP), manages the fishery resource of the state, including those within the reservoir. The nationally renowned wild trout populations, unique species composition, easy accessibility by vehicles, developed campgrounds, boat launch facilities, and scenic qualities result in heavy angling pressure. The lake is therefore stocked to support harvestable populations under heavy use.

The reservoir provides fisheries for introduced rainbow and brown trout, native burbot, and mountain whitefish. Occasionally westslope cutthroat trout and brook trout have been found. Non-game species in the reservoir include native white and longnose sucker and non-native common carp and the redside shiner (discovered in 1998). The rainbow trout is stocked by MTFWP with annual additions of hatchery fish and the other species are wild and self-sustaining.



MFWP Fish Trap Facility on Red Rock River

Rainbow stocking is generally done in early June to take advantage of a favorable thermal regime and the growth phase of the claoceran zooplankton community. In general, the population size trends with the storage and surface area of the reservoir. Since management strategy of the rainbow trout stocking program shifted to the use of wild strain DeSmet and Eagle Lake strain spawning stocks, they have been monitored in the Red Rock River for migration and spawning. The Clark Canyon Eagle Lake rainbow trout population became an effective wild brood source of fertilized eggs for rearing as over-wintered yearlings in 1995. These fertilized eggs are used for fish stocking in Clark Canyon and in other locations. Since the program began, Clark Canyon has provided 300,000 - 500,00 fertilized eggs annually (Oswald 2002a).

The Beaverhead River, just below the dam, supports a variable population of brown and rainbow trout with brown trout being the dominant species. The limited rainbow trout population is primarily found between Clark Canyon Dam and Dillon, Montana. Angler use of the river has been so great that the MTFWP instituted regulations to limit "float parties."

2.2.6 Threatened, Endangered and Sensitive Species

In this document, Threatened and Endangered (T&E) species include only species listed by the USFWS as threatened and endangered, under the ESA. U.S. Fish and Wildlife were informally consulted to obtain information on known occurrences of T&E species (US Fish and Wildlife 2003).

Wildlife Species

Bald Eagle (Haliaeetus leucocephalus) is a federally listed threatened species. The Fish and Wildlife Service reclassified the bald eagle (Haliaeetus leucocephalus) from endangered to threatened in the lower 48 States in 1995. The bald eagle is a bird of aquatic ecosystems (Gerrard and Bortolotti 1988). It frequents estuaries, large lakes, reservoirs, major rivers, and some seacoast habitats. However, such areas must have an adequate food base, perching areas, and nesting sites to support bald eagles. In winter, bald eagles often congregate at specific wintering sites that are generally close to open water and that offer good perch trees and night roosts. Bald eagle habitats encompass both public and private lands. Bald Eagles build large stick nests lined with soft materials such as grass, leaves, and Spanish moss. Nests are used for several years by the same pair of eagles, with the birds adding materials each year. Nests are often very large, measuring 6 feet across and weighing hundreds of pounds. Young eagles can fly in 11 to 12 weeks, but the parents continue to feed them for 4 to 6 more weeks while they learn to hunt. There are known bald eagles nests within two miles of the reservoir on both Red Rock River and Beaverhead River. Rock outcrops at Barrett Dam may provide suitable nest sites for the eagle. The rivers and in the reservoir provide foraging habitat for the eagle. Foraging in the reservoir is particularly popular in the winter for many individual birds. In general, the population has grown significantly in the last couple decades and cottonwoods along the streams are good potential nesting habitat (personal conversation with J. Rosco, Bureau of Land Management, Dillon Mt. 4/18/2003).

Other Wildlife Species

The Gray wolf, Canada lynx, and grizzly bear are also federally listed species and are listed as present in Beaverhead County but do not have suitable habitat in the project area (US Fish and Wildlife 2003). Therefore, these species will not be discussed further.

Plants Species

<u>Ute Ladies' Tresses (Spiranthes dilufialis</u>) is federally listed as a threatened species. It is a perennial orchid that arises from tuberous thickened roots and flowers in August-early September. It is known in braided streams with wetlands and swales in broad, open valleys, and at habitat margins with calcareous carbonate accumulation (MNHP 2002). *S. dilufialis* is listed as present in Beaverhead County but is not known in the project area (MNHP 2003). However, there may be potential habitat in the riparian areas adjacent to the reservoir and Beaverhead River.

2.3 Aesthetic Resources

This section describes the aesthetic environment of the study area, focusing on two primary aesthetic components: noise levels and the visual quality of the area. Although there are manmade features present and noise sources audible at both Clark Canyon Reservoir and Barretts Diversion Dam, each of these areas can afford visitors a general sense of solitude. The sections below describe the existing noise environmental and the visual character of the project area.

2.3.1 Noise Characteristics of the Project Area

Noise can be defined as the intensity, duration, and character of sounds from any and all sources. The discussion herein provides the general context of the existing noise environment within the study area. Measurements of actual noise levels within the study area have not been collected, and the discussion is based on observation and qualitative consideration of noise characteristics within the study area.



<u>Noise Sources</u> - In general, ambient noise levels at Clark Canyon Reservoir and Barretts Diversion Dam are consistent with rural, open space areas. Background noise levels are relatively low in most areas, with the notable exception of areas adjacent to roadways, including I-15 and SR 324, to the east and north of Clark Canyon Reservoir, and the Union Pacific railway, parallel to I-15 (see Section 2.7, Transportation Resources, for more detail on the transportation network within the study area).

I-15 experiences moderate volumes of truck and

passenger vehicle traffic, which create a relatively continuous noise source. SR 324 and other area roadways experience much lower traffic volumes and vehicle speeds are lower, resulting in lower and less consistent noise levels.

Barretts Diversion Dam is located on the northern end of a relatively narrow canyon through which I-15, the Union Pacific railway, and the Beaverhead River pass. Noise produced by vehicle travel along I-15 is audible at Barretts Diversion Dam; however, the contour relief provided by the elevated Union Pacific railway grade along with nearby trees may serve as a buffer, partially deflecting the noise generated along I-15 prior to its reaching Barretts Diversion Dam.

Other notable noise sources that contribute to the noise environment of the study area vary depending upon time of day, weather, and season, and include those associated with motorized recreational activities on and adjacent to Clark Canyon Reservoir, visitor activities at day-use and overnight campground areas, vehicular noise on area access roads, motorized water craft, and snowmobile traffic during the wintertime.

Of all the noise sources within the study area, motorized recreational vehicles (Off-Road vehicles (ORV) and watercraft in the summer and snowmobiles in the winter) may be the most prevalent. Noise from personal watercraft (PWC) and other motorized boats is reflected off the water and, depending upon weather conditions, can be heard at locations far from the source. In addition, because sounds levels from PWCs and motorized boats are highly variable as a result of engine revving (as opposed to more constant from a source such as pump or airplane) the noise tends to be more noticeable. In the wintertime, snowmobiles produce noise that is distinctly audible, especially in the context of relatively quiet winter days and nights.



<u>Sensitive Receptors</u> - The effects of noise on the aesthetic quality of an area is dependent upon factors including the presence of human receptors and the sensitivity of such receptors. Participants in noise-generating activities are considered to be less sensitive to noise. Conversely, visitors or residents not participating in activities which create significant noise levels may be seeking solitude, and are therefore, more adversely affected by the presence of significant continuous or variable noise. It should be noted, that individual recreationists within the study area, might be considered as sensitive or less sensitive, depending upon time of day and the activities in which they are participating.

Evening, nighttime, and early morning hours are the times of day that most area visitors and residents are most sensitive to noise. Thus, sensitive receptors are primarily considered to be campers present in the study area overnight, as well as some anglers.

Sensitive residential receptor locations within the area include residences located adjacent to Barretts Diversion Dam, on the east side of the Beaverhead River and to the east of the existing Barretts

Diversion Dam day-use shelter. A small number of residences located east of the I-15/SR 324 interchange (I-15 Exit 44) may not be considered sensitive receptors. Campgrounds and day-use areas adjacent to Clark Canyon Reservoir and at Barretts Diversion Dam (see Site Map) and a private campground located east of Exit 44 are also considered to be areas that are sensitive to noise associated with highway and road traffic.

2.3.2 Visual Resources and Quality of the Project Area

Specific scenic resources and the overall visual setting of the study area are recognized as an important component for recreational enjoyment of visitors, as well as to residents and travelers passing through the area. Due to the general openness afforded by gently sloping terrain that surrounds much of Clark Canyon Reservoir, the reservoir is highly visible from adjacent lands, and is a dominant feature of the landscape to travelers on I-15. Barretts Diversion Dam is less visible from surrounding areas and is located at the base of a steep canyon wall that is the most dominant feature. As such, the park is most visible primarily to visitors at the park and residents living east of the park. The following sections describe each of these areas in more detail.

<u>Clark Canyon Reservoir</u> - Although Clark Canyon Reservoir is man-made, the reservoir offers a some-what natural appearance, visible from surrounding shoreline areas and travelers on I-15. The reservoir is situated in a broad valley at the confluence of the Red Rock River and Horse Prairie Creek, and maintains a roughly triangular shape. The southwest shoreline is characterized by relatively gently rising hillsides that extend gradually upward and away from the reservoir whereas the northern shoreline is somewhat steeper and rises more abruptly. The eastern shoreline has a linear appearance, accentuated by I-15 and the Union Pacific rail tracks, which run parallel to the eastern shoreline of the reservoir.

The southern and western shorelines provide more of a marshy appearance, primarily in association with the stream deltas of Red Rock River and Horse Prairie Creek. Other than these two stream delta areas, shoreline areas generally consist of grasses, shrubs, and barren areas with compact or sandy soils that slope to the water's edge. Surrounding terrain is generally open, with low vegetation consisting of grasses, shrubs, and brush, broken up by occasional clumps of taller and denser vegetative cover.



Two visually dominant features in the northern end of the reservoir are Clark Canyon Dam and Armstead Island. Clark Canyon Dam is an earth and rock filled dam. The top of the dam serves as the crossing for SR 324. Armstead Island is dominant in the view of the reservoir from the Lewis and Clark Overlook, as well as most other nearby locations. The northern and western portions of the sparsely vegetated island rise abruptly from the surface of the reservoir, while the eastern and southern shores of the island have a more gradual slope.

The reservoir surface elevation fluctuates seasonally. During periods when reservoir levels are low there is a significantly reduced water surface area and a wide shoreline approach to the reservoir, especially along the gently sloping western and southern shorelines.

During wintertime, the reservoir surface freezes and is often snow-covered. In the past, ice fishing has been a common wintertime activity at the reservoir. Temporary ice fishing structures are placed on the reservoir and are visible from surrounding areas, including motorists on I-15. The structures provide contrast to the otherwise barren, ice-covered reservoir surface, and can be a common element of wintertime in the region.

<u>Barretts Diversion Dam</u>- Barretts Diversion Dam is located on the northern end of a relatively narrow canyon through which I-15, the Union Pacific Railroad, and the Beaverhead River pass. The park's dominant features include a gravel parking area, turf areas on the Beaverhead River which bisects the park, restroom facilities, signage, and a day-use group shelter on the east side of the river. Two small bridges are situated within the park, one for pedestrian access to the portion of the park on the east side of the river, and the other for vehicle access to the private residence on the east of the river.

Immediately east of the park and the Barretts Diversion Dam, steep canyon walls rise abruptly and provide a dominant, natural backdrop to the park's east and north facing viewers. With the exception of compacted earth parking areas, most of the park is covered by maintained grass, and numerous large deciduous trees provide canopy cover over much of the park area during the spring and summer months.



Recreational Facilities - Other than the dam, spillway, and outlet facilities, the developed recreation facilities surrounding Clark Canyon Reservoir are the most visible man-made structures other than access roads, the railroad, and highways. The most visually dominant recreation areas are Beaverhead Campground located east of the dam, the marina (which is currently unused, but contains structures and signage), and Camp These facilities contain access Fortunate Overlook. roads. delineated campsites and parking areas. restrooms, open-air shelters, and other facilities that are visually evident to users and travelers passing by.

Although some of the facilities are readily visible to most visitors and in mid-ground views of travelers on I-15 and SR 324, the facilities are not considered dominant in comparison to the scale of the overall surrounding view sheds. Other Clark Canyon Reservoir sites contain similar facilities, but generally have fewer structures and are situated in areas less visible from area visitors and residents.

Minimal night lighting is installed and operational at Barretts Diversion Dam, Beaverhead Campground and the marina. Night lighting is provided at these facilities for the purposes of safety, security, and user convenience. Lighting is limited to parking areas and buildings to provide some

visibility during nighttime. Although such lighting can be visible from surrounding areas, it does not contribute a significantly intrusive degree of light or glare to the project area. Other sources of light associated with the project area are from vehicles on area roadways and within campgrounds, lighting associated with the few private residences and businesses adjacent to Clark Canyon Reservoir and Barretts Diversion Dam, also campfires and individual camp lighting (e.g., lanterns, motor home lights, etc.).



<u>Wildlife Viewing Areas</u> - Wildlife Viewing Areas have been designated within the study area. Wildlife viewing areas include the developed bird watching trail below the dam. Other wildlife viewing areas include the low lands around the mouth of Horse Prairie Creek and Red Rock River. Antelope can frequently be seen on the southern hillsides above the reservoir. Moose have been seen in the Horse Prairie Creek and Red Rock River delta areas, while shorebirds and other waterfowl are abundant throughout the area.

<u>Scenic Byways</u> - Big Sheep Creek Back Country Scenic Byway is located south and west of Clark Canyon Reservoir. The roadway intersects with SR 324 west of Clark Canyon Reservoir, heads generally south then east, passing south of the Big Sheep Back Country, before turning northeast to I-15 near the town of Dell. The roadway is not considered to be within the study area; however, it is discussed briefly here due to its proximity. According to the Montana Department of Transportation (MDT):

"The road is mostly two-lane gravel with a few side roads that lead to the foot of the Rocky Mountains that provide many opportunities for solitude and exploration. Once past the canyons of the Tendoy Mountains, the traveler may pass only two or three cars along the entire route. Passing beneath the high rock cliffs of Big Sheep Canyon, one can look down into clear, deep pools of spring-fed Big Sheep Creek. Perhaps nowhere else in America can one see so many trout without leaving the car. Bighorn sheep and deer are a common sight in the evenings. The Byway provides one with the opportunity to see this country as it was experienced by prehistoric Indians and the first mountain men. The Rocky Mountains have hidden this southernmost point of Montana well, and it remains unchanged for those willing to explore the back country." (MDT, 2003)

Motorists accessing the Big Sheep Creek Back Country Byway via SR 324 from I-15 cross Clark Canyon Dam and traverse the northern portion of the study area. Because the scenic byway is west of the study area the reservoir lands, the facilities are out of the scenic road's view shed.

<u>Wild and Scenic Rivers</u> - A segment of the Beaverhead River between Clark Canyon Reservoir and Barretts Diversion Dam has been evaluated for potential eligibility and possible inclusion in the National Wild and Scenic River system. The evaluation concluded that this segment would be eligible with a classification of "Recreation", as presented in a recent report issued by the Montana BLM, Dillon Field Office entitled, "Final Report on Wild and Scenic River Eligibility Determinations" released in July 2002. The segment begins at the I-15 bridge at Pipe Organ Rock and ends where it leaves BLM land near the Dalys exit on I-15. This segment of the Beaverhead is considered "outstandingly remarkable" for recreation, fish, and historic values. This section of the

Beaverhead River is not within lands under Reclamation's jurisdiction and is located outside the project area.

2.4 Cultural Resources

2.4.1 Introduction

Since the early 1950's both intensive and nonintensive cultural resource surveys have been conducted at the Clark Canyon Reservoir (Deaver and Mortrer 1981) (Fenenga 1951). These studies have been carried out to comply with one or more Federal historic preservation laws and regulations. These laws and regulations direct Federal Agencies to manage cultural resources and set forth a policy to provide leadership in the preservation of the prehistoric and historic resources of the United States.

Federal historic preservation laws are designed to protect cultural resources for future generations and to promote scientific study of those resources. Studying the past provides insight into how humans utilized the environment in which they lived, and can provide keys to future changes.

In 1951 an Appraisal of the Archaeological and Paleontological Resources of the Jefferson River Basin, Montana was completed by the Missouri River Basin Survey of the Smithsonian Institution (Fenenga 1951). This survey included what is now the Clark Canyon Reservoir Area. The cultural resource survey, upon which this report was based, is very preliminary, and only a few late prehistoric sites were reported.

Following passage of the National Historic Preservation Act in 1966, cultural resource inventories in the Clark Canyon increased as Reclamation responded to the new mandates found in the legislation. In 1979, a cultural resource survey was conducted of the recreation area at Barretts, and at three recreation areas at Clark Canyon Reservoir. A cultural resource survey for the entire reservoir area above the water line was undertaken with the report completed in 1981. Both of these surveys have been useful for management of cultural resources although neither were performed to current standards making additional survey work necessary before specific projects are undertaken.

2.4.2 Prehistoric Period Resources

Research conducted in the area has revealed prehistoric sites and use areas that demonstrate extensive Native American utilization of the region. Artifact scatters and quarry sites show the area was used for stone tool production, bison kills and butchering sites demonstrate meat procurement, and camps with fire hearths and grinding stones demonstrate food preparation (Deaver 1981). Although none of the prehistoric sites at Clark Canyon Reservoir have been nominated to the National Register of Historic Places some may be eligible since they may be likely to yield information important in prehistory or history.



Beaver Slide

2.4.3 Historic Period Resources

The historic period at Clark Canyon Reservoir dawned in 1805 when the Lewis and Clark Expedition entered the area (Malone 1976). Their meeting with the Lemhi Shoshoni Indians at "Camp Fortunate" at a location now under the waters of Clark Canyon Reservoir was an important factor in the success of the expedition. The Lemhi Shoshoni Indians were not the only tribe known to have utilized the area; research has reported usage by the Bannock, Flathead, Pend Oreille, Nez Perce and Blackfeet. Even as the Lewis and Clark expedition returned to St. Louis, trapping beaver and fur trading was becoming key to the economy of the area. Although no major fur trading posts were developed in the Clark Canyon Area, camps of traders were often located in the broad valleys of southwest Montana. Members of the tribes found in the area trapped furs and traded as well as others from outside the region.

The broad valleys of the Beaverhead and Red Rock Rivers, along with Horse Prairie Creek have been used as travel routes for thousands of years. One likely prehistoric trail, marked by a cairn line bypasses a location that would be difficult to travel during many times of the year. Following the establishment of the Oregon Trail in Idaho, travel into the area was north over Monida Pass, and then down the Red Rock River through the study area. This route, followed portions of the prehistoric trail, was followed not only by early travelers but stage and freight lines, and then the railroad.

Discovery of gold by John White's party in 1862 started the gold rush to Montana. Bannack and Virginia City, located northwest and northeast of Clark Canyon became leading mining towns. Support systems including merchants, traders, stage transportation companies and saloons developed. By 1869, Corinne, Utah on the Union Pacific Transcontinental Railroad was the center for stage and freight traffic north to Montana. The trail was established from Corinne, Utah through Monida Pass, down the Red Rock River and on to Bannack, Virginia City, and other points north. Travel along this route brought many into the area. In the 1860's cattle and sheep were brought from California and Oregon into Montana utilizing these trails. The Utah and Northern Railroad built north across Monida Pass in March of 1880 and Dillon was founded as a terminus town (Malone 1976).

The Gilmore & Pittsburgh Railroad was constructed from the mainline of the Utah and Northern up Horse-Prairie Creek to the mines at Salmon and Gilmore, Idaho. A community named Armsted, Montana developed at the junction of the two railroads. Portions of the grade were so steep, that the Gilmore & Pittsburgh gained the nickname "Get Out and Push". The G, O & P was not profitable, and the tracks were removed in the early 1940's. The post office remained open at Armstead until 1962 (Anon 1990).

Construction of the Clark Canyon Dam was completed in 1964. With the filling of the Reservoir numerous changes occurred to the cultural landscape in the immediate area of the project. The railroad was relocated to higher ground, as were the highways. Farms and ranches that were located in the valley bottom and the town of Armstead were removed. Few historic remains are located on Reclamation land adjacent to the Reservoir, and none are on the register of historic places.

2.4.4 Future Heritage Resources Focus

The Federal Government is required by law and regulations to protect and preserve significant cultural resources. To this end, Reclamation's undertakings are subject to compliance required by the National Historic Preservation Act and its regulations found in 36 CFR Part 800. These mandate require that the Government consider the effects of its action or undertaking on prehistoric and historic resources before implementing those actions. Reclamation routinely consults with the Montana State Historic Preservation Officer (SHPO) and other interested parties before undertakings take place. The objective of the compliance process is to reduce or mitigate negative impacts to cultural resources.

2.4.5 Paleontological Resources

Like Cultural Resources, Paleontological Resources are fragile and nonrenewable. At Clark Canyon Reservoir, the geology is such that the presence of such resources is unlikely, and none have been reported.

2.4.6 Indian Trust Assets

Indian Trust Assets (ITAs) are legal interests in property held in trust by the United States for Indian Tribes or individuals. Examples of things that may be ITAs are lands, minerals, hunting and fishing rights, and water rights. The United States has an Indian trust responsibility to protect and maintain rights reserved by or granted to Indian Tribes or Indian individuals by treaties, statutes, and Executive orders; these right are sometimes further interpreted through court decisions and regulations. The trust responsibility required that all Federal agencies, including Reclamation, take all actions reasonably necessary to protect trust assets.

2.4.7 Indian Sacred Sites

Executive Order # 13007 defines Indian sacred sites as "...any specific, discrete, narrowly delineated location on Federal land that is identified by an Indian Tribe or Indian individual determined to be an appropriately authoritative representative of an Indian religion, as sacred by virtue of its established religious significance to, or ceremonial use by, an Indian religion: provided that the Tribe or appropriately authoritative representative of an Indian religion has informed the agency of the existence of such a site." Federal agencies are required, to the extent practicable, to accommodate access to, and ceremonial use of, Indian sacred sites by Indian religious practitioners and to avoid adversely affecting the physical integrity of such sites.



2.5 Land Use and Management

Approximately 70 percent of lands within Beaverhead County are under ownership or administration by public (federal and state) agencies. The largest portion (over 40 percent) is owned or managed by the U.S. Forest Service (Forest Service) and approximately 18 percent is owned/managed by the U.S. BLM. Lands under the jurisdiction of Reclamation comprise 4,747 acres, or approximately 0.1 percent of the total land area of Beaverhead County. A summary of land ownership within the county is provided in Table 2.5-1.

Table 2.5-1 Land Ownership/Administration in Beaverhead County						
Ownership/Administration Acres Percent of County Total						
Reclamation	4,747	0.1				
BLM	662,940	18.7				
U.S. Forest Service	1,442,682	40.7				
U.S. Fish and Wildlife Service	39,229	1.1				
U.S. National Park Service	675	< 0.1				
U.S. Agricultural Research	15,508	0.4				
Service						
Montana FWP	28,845	0.8				
Montana DNRC	1,819	0.1				
Montana State Lands	322,015	9.1				
Private	1,007,200	28.4				
Other	19,433	0.6				
Total	3,545,093	100				

2.5.1 Clark Canyon Reservoir and Adjacent Lands

Reclamation has management authority and jurisdiction over lands at Clark Canyon Reservoir and those that comprise Barretts Diversion Dam, see Land Ownership Map (Figure 2.5-1). A total of 800 acres of the Reclamation land surrounding Clark Canyon Reservoir is included within a land withdrawal granted by the BLM to Reclamation.¹ Lands surrounding those under Reclamation's jurisdiction include a combination of lands in private ownership and public ownership, including lands under the jurisdiction the BLM, U.S. Forest Service, and Montana Fish, Wildlife and Parks.

The lands and resources at Clark Canyon Reservoir are publicly owned and managed through a coordinated effort between several entities that have different management responsibilities. Reclamation maintains primary jurisdiction of the lands within the study area (Clark Canyon Reservoir and Barretts Diversion Dam) and is responsible for area resource management. Montana Fish, Wildlife and Parks is responsible for managing for fish and wildlife resources at Clark Canyon Reservoir and oversees the regulation of fisheries and performs fish stocking to enhance the reservoir fishery. Reclamation contracts with the Beaverhead County Weed District to treat noxious weeds at Clark Canyon Reservoir and Barretts Diversion Dam. The East Bench Irrigation District,

¹ A withdrawal is a formal action that sets aside, withholds, or reserves Federal lands by administrative order or statue for public purposes.

with over site from Reclamation, operates and manages the reservoir water operations and Clark Canyon Dam.

Reclamation's policies govern the use of lands within its jurisdiction and affect the way lands are managed in the future. Non-Reclamation land-use activities are authorized and managed under outgrants such as license agreements, lease agreements, right-of-way easements, special use permits, and other legal and binding contracts. There are numerous such agreements currently in place and include use permits for utility right-of-way, road maintenance, and fire suppression. In the past, Reclamation has entered into cooperative agreements for law enforcement.

Each land use for example authorization is subject to specific terms and conditions covering the use of the Federal estate. Lessees are restricted from conveying their permitted use to another party without the approval of Reclamation. Reclamation's Montana Area Office (MTAO) has copies of all authorized outgrant documents dealing with third-party use of Reclamation project purposes; they are to be, for the most part, temporary in nature and contain restrictive language that protects present and future Federal land interests.

A portion of the Land Use Authorization Directives and Standards states that Reclamation will prohibit any new exclusive/semiprivate use of Reclamation land unless directed otherwise in specific authorizing legislation and that existing private/semiprivate use will be eliminated when the use authorization expires unless a formal planning process determines that there is a significant public need and benefit for the exclusive private/semiprivate use and the land is not needed for other public or project purposes.



In addition, Reclamation issues permits for all improvements within easements, such as culverts and bridges, pipelines, and utilities. Special use permits are also considered for rock collecting, archeological investigation, airports, wells, mineral exploration and extraction (including sand and gravel), fire protection facilities, surface water use, and material storage. Pesticide application on Reclamation lands requires a plan and permit.

Periodically, Reclamation also evaluates its lands of possible future disposition. Pursuant to Section 204 of the Federal Land Policy and Management Act of October 21, 1976, Reclamation must review all its withdrawn lands to determine if they are needed for project purposes. Withdrawn lands not needed for project purposes should be returned to the public domain for administration by the Bureau of Land Management (BLM) or some other Federal agency that had original jurisdiction.

In the case of acquired lands, Reclamation must make a determination that the acquired lands are no longer needed for project purposes and are excess lands. These lands are reported as excess to the General Services Administration (GSA). GSA first offers the lands for sale to other Federal, State, or local public entities; if these entities do not want the lands, they are put up for sale through public bid at established fair market and competitive prices. Currently, all lands within Clark Canyon

Reservoir and Barretts Diversion Dam are needed for authorized project purposes and are not available for disposal.

2.5.2 Land Use Authorizations and Land Disposal

Land use authorizations include easements, leases, licenses, and permits which allow others to use Reclamation lands, facilities, and water surfaces. Land use authorizations are discretionary and must conform to the requirements contained in 43 CFR 429 and Reclamation Directives and standards for Land Use Authorizations (LND 08-01). Issuances of land use authorizations, require that Reclamation collect administrative fees and receive fair market value for the use of its lands.

Land use authorizations will not be issued when it is determined that the proposed use is incompatible with authorized project purposes, or where the proposed use poses health and safety concerns, results in unacceptable impacts to the environment, results in private exclusive uses, violates state, Federal, or local laws, regulations, ordinances, or zoning requirements, jeopardizes the interests of the United States, is an existing unauthorized use, the use will result in other adverse and unacceptable impacts, or where other alternatives are available. If a use authorization is granted, it is the policy of Reclamation to grant the least estate possible necessary to accommodate the intended use. Generally, this means that Reclamation will only issue a permit or a license, and will not issue leases or easements or other contractual documents that convey an interest in real property.

In regard to roads, telecommunication sites, microwave towers, transmission lines, and linear facilities, Reclamation will not issue any land use authorizations for these uses that does not clearly compliment the goals and recommendations contained in this plan. In the event electrical utilities are permitted they shall be buried, or if constructed above ground, they shall be constructed and only permitted if they do not interfere with the visual and aesthetic components of the landscape. All above ground utilities will provide measures of safety for the public and for wildlife species. All land use authorizations will be considered on a case-by-case basis, and issued at the sole discretion of Reclamation.

All reservoir lands are currently needed for project purposes and are not available for disposal thru sale, lease, or transfer to any other person, agency, or entity. In the event any of the lands within the reservoir are determined by Reclamation to be excess to project needs Reclamation will follow standard General Service Administration procedures to dispose of the lands. This process requires that any excess lands be first offered to other Federal or State agencies, then to local counties or municipalities, and lastly thru sale at public auction to the highest bidder.

Requests from outside parties, to use, project sand and gravel resources will not be allowed unless such use directly benefits Clark Canyon Reservoir or Barretts Diversion Dam and Park. Proposed oil and gas leasing/drilling may be permitted in accordance with BLM's mineral leasing program and subject to Reclamation's concurrence and subject to it's GP-135 special stipulations and form 3109-1 lease stipulations.



2.5.3 Barretts Diversion Dam and Adjacent Lands

The area surrounding Barretts Diversion Dam is primarily in private ownership. There are also transportation rights-of-way and BLM lands to the west of the park. The U.S. Government, through the Bureau of Reclamation owns the Barrett Diversion Dam and surrounding lands that comprise Barretts Diversion Dam. The East Bench Irrigation District operates and maintains the East Bench Canal that delivers water to approximately 23,000 acres held in various private ownerships north of Barretts Diversion Dam (see Figure 2.5-2).

2.5.4 Land Management Planning

In addition to Reclamation's preparation and implementation of a Resource Management Plan for Clark Canyon Reservoir and Barretts Diversion Dam, other federal and state agencies maintain management plans, which guide their land, and resource management planning. Identifying the plans other area land management agencies provides important context for Reclamation's management activities. Coordinating land management activities with adjoining land management agencies help to reduce inconsistencies in uses and land management objectives and can help to reduce costs and minimize potential areas of conflict between uses.

2.5.5 Other Resource Management Planning



BLM is currently updating and preparing a Resource Management Plan for the management of lands and resources within the jurisdiction of the BLM's Dillon Field Office. These lands include those adjacent to Reclamation lands surrounding Clark Canvon Reservoir as well as lands located within the Clark Canyon Reservoir watershed, and therefore the decisions and direction taken by BLM in management of these public lands are an important consideration for Reclamation in its management decision. In January of 2003, BLM released baseline information collected

as part of its RMP development and NEPA compliance effort. Since that time, BLM and cooperating agencies have hosted a series of public workshops to gather public input concerning the project. BLM released the Draft RMP and EIS for public comment in March 2004. BLM anticipates completion and release of an approved RMP by August of 2005.



2.5.6 Montana Statewide Comprehensive Outdoor Recreation Plan (SCORP)

The State of Montana's Statewide Comprehensive Outdoor Recreation Plan (SCORP) identified needs and opportunities for outdoor activities within the state during the 2003 through 2007 planning period. Although the SCORP does not contain provisions or requirements that are directly applicable to Reclamation's management of resources and recreation at Clark Canyon Reservoir and Barretts Diversion Dam. The findings and objectives identified in the SCORP are relevant to Reclamation's management planning efforts. In as much as efforts to contribute to statewide recreation objectives could influence Reclamation's management decisions. The two most relevant findings are the need to enhance access for water-based recreation activities and to improve access for wildlife-based recreation activities.

2.5.7 Montana Fish, Wildlife and Parks

The Montana Fish, Wildlife and Parks (MFWP) manages fish and wildlife activities within the state, including certain fishing and boating activities on the Beaverhead River below Clark Canyon Dam. MFWP is currently developing a river recreation management program, planned for completion in 2005. To develop the program, MFWP intends to work with the public on developing reasonable and equitable proposals for funding a river recreation management program to carry out the state's final river recreation management plan. The final river recreation management policy the MFWP Commission adopts will reflect comments from the public, the recommendations of the River Recreation Advisory Council (RRAC), and input from MFWP river managers and others. The river recreation management policy will guide decision-making and management actions for individual rivers around the state.

In order to learn more about the interests and values of the public, MFWP appointed the RRAC in August of 2002. The 22 council members represent anglers, boaters, outfitters, landowners, the Montana state legislature, the tourism industry, and the MFWP Commission. The RRAC developed recommendations for addressing the public's concerns about overcrowding and other social issues on rivers in Montana. The RRAC presented its recommendations to the MFWP Commission in July of 2003 and MFWP will use them to develop a river recreation management policy and program. For lands surrounding Clark Canyon Reservoir, MFWP is currently in the process of modifying its hunting regulations applicable to these lands. The new regulations will only allow the use of archery, shotgun, handgun, and black powder for hunting any game animal.

2.6 Recreation and Visitation



The primary land use in the Project area is for public recreation. The area is especially suited to the pursuit of unconfined and generally primitive recreation opportunities and activities.

Recreation activities in the Project area consist of land and water-based activities that take place primarily from May 1 through Labor Day weekend in early

Draft Resource Management Plan For Clark Canyon Reservoir and Barretts Diversion Dam

September. In winter, there can also be local weekend use associated with ice fishing when conditions are appropriate. Most of the recreation users are from eastern Idaho and south central Montana. Residents of these areas are able to reach Clark Canyon Reservoir within 2.5 hours driving time (defined as the area of influence).

Clark Canyon Reservoir provides regional camping and water based recreation opportunities for the communities of Dillon, Anaconda, Butte and Idaho Falls as well as other Idaho and Montana residents. In addition, other out-of-state residents who are either passing through the area or are in the area to fish also patronize the facilities. It is estimated that approximately 59,000 people visit the area during the summer season.

The activities most visitors engage in include fishing, boating, camping, picnicking, swimming, and wildlife viewing.

2.6.1 Fishing

Clark Canyon Reservoir is the largest and most accessible reservoir in this part of south central Montana and provides a popular trout fishery. Anglers primarily catch burbot (ling), brown trout, and rainbow trout. Trout are the most important recreational and sustenance species. Until recently anglers were allowed 5 trout and 10 burbot (ling) per outing, but beginning in 2002 new rules by MFWP limited the catch to no more than 2 trout and 2 ling per angler day. The new regulations were put in place mostly in response to low reservoir conditions. The Beaverhead River directly below Clark Canyon Dam is a Blue Ribbon Trout fishery. Angler use is relatively constant from May through October with the peak use in June and July.

2.6.2 Boating

Motorized boats are the principle means to access and enjoy many opportunities available at Clark Canyon Reservoir. Motorboats support such popular activities as fishing, sightseeing, and boat-in camping. There are also some minor amounts of waterskiing, tubing, sailboarding, and personal watercraft uses (i.e. jet skiing). The reservoir is four miles in length and three miles in width and provides 17 miles of shoreline. The highest concentration of boating activities occurs between Armstead Island and the dam. The reservoir is situated in a high mountain valley that is often exposed to high winds, which often hamper boating activities. The few coves found on the reservoir margins become important shelters when water levels are high enough to cover and fill the cove bottoms.

Primary boat launch facilities at Clark Canyon Reservoir are located at the Beaverhead Campground and the marina. Unmaintained launch areas that can be used for smaller boats and PWC are available at certain other day-use and campground areas. The marina launch has not been used during the past two seasons, and the Beaverhead boat ramp has been the primary launch site. The reservoir can experience high winds during certain periods which can make boat launching difficult and limit boat-launching opportunities.

2.6.3 Off-Road Vehicle (ORV) Use



All Reclamation lands in the Project area are closed to motorized travel except designated roads. If ORV uses were to be accommodated, a plan would have to be developed and designated areas established as required by 43 CRF 420. However, the land areas are small, the low-lying vegetation is fragile, and most areas are used primarily for water based recreation uses. In spite of such closures, unauthorized ORV use occurs around several recreation sites. One of the areas in which ORV use has been extensive is along the shoreline between Lone Tree Campground and the old highway near the Red Rock River.

Reclamation has been erecting jackleg fencing (crossed wood pole fences) to contain ORV use and this has been effective where it has been employed. On adjacent lands administered by the BLM, management designations generally preclude ORV use, which is also limited to existing roads and trails only.

2.6.4 Hiking

Other than the Watchable Wildlife Trail below the dam, there are no other constructed trails in the vicinity of Clark Canyon Reservoir. However, boat-in and other overnight campers on Clark Canyon Reservoir often choose hiking as their primary day-use activity. Camping opportunities on Armstead Island, at Hap Hawkins, Lone Tree and West Cameahwait Campgrounds offer excellent hiking opportunities into highly scenic and diverse landscapes.

2.6.5 Recreation Sites and Facilities

Recreation sites and facilities within the Project area included developed and primitive day and overnight use areas, boat ramp, a marina, and popular fishing and group day use sites.

Following is an overview of the public recreation sites and facilities that are managed by Reclamation along the Beaverhead River and at Clark Canyon Reservoir. This includes developed, semi-developed, undeveloped, and random (dispersed) recreation sites.

2.6.6 Dispersed Sites

Dispersed sites are recreation areas defined as sites with no formal names and have an absence of facilities (or minimal facilities) around the reservoir. The dispersed recreation sites are an important complement to the developed sites. By using the many miles of undeveloped shoreline around the reservoir, visitors can gain a feeling of solitude just a short distance from developed sites. Due to the diversity of natural features, the dispersed recreation areas provide opportunities for wildlife observation, nature study, hiking, boating, and fishing. This helps to distribute use, reduce potential

user conflicts, and allow otherwise incompatible uses to take place at the same time. Dispersed area use tends to be less concentrated, less noisy, and less intense compared to use at the developed sites.

However, lack of sanitary facilities, trash receptacles, and designated parking areas in the dispersed areas has contributed to a decline in the quality of vegetative cover and natural soil conditions. Visitors sometimes leave trash, use the rock or bush as an open-air restroom, and drive and park wherever possible and convenient.

Nevertheless, the dispersed recreation areas around the reservoir are important components of the overall recreation spectrum. Water-based uses such as waterskiing, wind surfing, power boating, sailing, and boat fishing occur throughout the lake area, but are supported and enhanced by a number of dispersed and developed recreation sites along the shoreline. The quality of the recreation experience and natural resources of the area could be enhanced if minimum basic facilities are provided at dispersed recreation sites. Well-used dispersed sites include the Sacajawea Day Use. The Sacajawea Day Use area is a beachfront



cove area with roads and has several road spurs that led to private cabins that stood in this area at one time. The site is situated off SR 324 between Camp Fortunate and HorsePrairie Campground. Dillon Beach is a popular beach area that is accessed through Cameahwait Campground and is a popular beach area with college students in Dillon; however, there are no facilities at the site.

2.6.7 Developed Sites

Recreation facilities currently provided at Clark Canyon Reservoir and Barretts Diversion Dam include 15 developed or formally recognized recreation facilities as summarized in Table 2.6-1.

Each of the developed sites at Clark Canyon Reservoir provides unique settings that offer visitors different options and features. The following description provides some background and characteristics of each site that are important considerations in the planning process. As part of the facility descriptions, the accessibility of the facility is described to provide a baseline for improvements that might be needed under the Americans with Disabilities Act of 1990 (Public Law 101-336). The ADA is a comprehensive civil rights law that prohibits discrimination on the basis of disability. It also requires that newly constructed and altered state and local government buildings and facilities, places of public accommodation, and commercial facilities be readily accessible to and usable by individuals with disabilities.

Name	# Picnic Sites	# Camp Sites	Group Shelters	Boat Ramp	Potable Water	# Toilet	Other
Barretts Diversion Dam	Shared	22	1	1	VES	<u> </u>	Tent camping
	Sharea	22	1	1	1125		Tent camping
Beaverhead River Fishing Access	Shared	5	-	1	YES	2	Wildlife trail &
				(unimproved)			interpretive signs
Buffalo Bridge Fishing Access	None	2	-	1 (unimproved)	NO	1	Remote parking
Beaverhead Marina	Shared	36	-	1	YES	3	RV hookup sites
Lewis & Clark Day Use Area	4	None	-	-	YES	1	Sites have wind shelters
Beaverhead Campground	Shared	22	-	2	YES	3	1 ramp is for low water. End of ramp elevation is 5512
Camp Fortunate Overlook	1	-	1	-	NO	1	Interpretive facilities
Horse Prairie Campground	Shared	17	-	1	YES	4	West & East units; most sites have shelters
Cameahwait Campground	Shared	11	-	-	YES	2	4 shelters
West Cameahwait Campground	Shared	2	2	-	YES	2	Primarily a group site
Hap Hawkins Campground	Shared	2	1	-	YES	1	Primarily a group site; has outdoor amphitheatre
Lone Tree Campground	Shared	20	-	1 (rough)	YES	4	Most sites have shelters
South Shore Fishing Access	None	-	-	-	NO	-	No formal parking
Red Rock Fishing Access	None	-	-	-	NO	-	Small gravel parking lot
Armstead Island Campground	Shared	2	-	-	NO	1	Boat-in only

Table 2.6-1. Clark Canyon Reservoir Recreation Facilities

<u>Barretts Diversion Dam</u>– Barretts Diversion Dam is a popular day use and overnight camping area situated on approximately 36 acres adjacent to the Barretts Diversion Dam. The site also has historic values since it is a camp location specifically noted in the journals of the Lewis and Clark Expedition of 1805-06. A large rock outcrop on the east bank, also noted in the Lewis and Clark journals, provides a dramatic drop back to the river and park site on the west bank. The park is set up to accommodate RV users and has a boat ramp that serves as a popular take out point for anglers using small drift boats. The park is used extensively by Dillon residents who often come in

organized groups. Demand is high for summer weekend use and sometimes groups end up competing for the single group shelter, or pavilion, on the same day. In recent years, the park has become more popular for outdoor weddings, business or civic organization picnics, and family reunions. The park entrance road and bridge are also used to provide access to a small ranch located directly adjacent to the park's southern boundary. Currently visitors using the group area must get special permission to drive vehicles across the bridge to access the pavilion. While many of the facilities are compliant with the Americans with Disabilities Act (ADA) standards, the pedestrian access to the facilities is not barrier free, and thus not fully ADA compliant.

<u>Beaverhead River Fishing Access</u> - This facility is situated directly below Clark Canyon Dam, and is often called the "campsite below the dam" or "below dam campsite". The gravel road leading down to the site is steep, somewhat narrow and has windy switchback turns. The facility is however well used by anglers and as a put in point for rafters. The site is used by both day users and campers, and the river setting is well vegetated and very appealing to campers and those who prefer a free-flowing river setting. A formal wildlife trail is part of the site's attraction and the trail is used by bird watchers and other wildlife enthusiasts as well as the campers and other day users. The wildlife trail has several interpretive signs and is listed in the Montana Fish, Wildlife and Park's "Watchable Wildlife Trails" program. The trail is managed by Reclamation.



<u>Buffalo Bridge Fishing Access</u> - The Buffalo Bridge Fishing Access site is not a formal designated or signed site but it is well known and used by anglers and commercial outfitters who launch their drift boats or other craft, serving as a "put-in" location. The site has a rough boat ramp and has limited space for boat trailer turn around and maneuvering. Additionally the parking spaces and launching lanes are not well delineated. Access and parking for the site is situated on the east side of I-15, adjacent to a private RV park. The access pathways to the toilet at the launch site are not currently accessible according to ADA standards.

<u>Beaverhead Marina</u> - The Beaverhead Marina facility has camping and boating facilities including campsites with full RV hookups along with a store and boat ramp. At this time only the campsites and boat ramp are open to the public as there is presently no concessionaire to run the marina facility. Reclamation would consider entering into an agreement with a concessionaire that would be able to provide gas, boating supplies, mooring slips, and some groceries as well as run the camping and day use facilities. However, because most operators in the past have had a difficult time making a profit, there has been little interest by new parties especially since upgrades are needed. At low lake levels the existing boat launch and dock area become too shallow for use by most watercraft.

<u>Lewis & Clark Day Use Area</u> - This site was recently converted day use only, as it was primarily a campground until the primary access road leading to Camp Fortunate Overlook was re-routed through the site. As a campground, the site was not well used due to lack of flat and level parking spots and lack of privacy and shelter, trees, grass or shrubs. Starting in 2003 the site was opened to

only day use and it is hoped the facilities will be better used for picnicking and other use by the visitors who are increasingly patronizing the Camp Fortunate Overlook facility.

<u>Beaverhead Campground</u> - This facility includes two boat ramps and a well-used campground. The facility is easily accessible off of I-15 and has the most trees and maintained (irrigated) lawn areas of any of the sites other than Barretts Diversion Dam. One boat ramp (adjacent to the dam) is especially suited for low water launching conditions, although it has a fairly steep launch lane. The other boat launch lane is more popular during periods of higher reservoir levels. The parking areas for each boat launch are not well defined and vehicle parking in vegetated areas has damage some of the ground cover and exposed native soils to erosion. The camping sites are in good condition but there are not very many level parking sites for large RV users.



<u>Camp Fortunate Overlook</u> - Camp Fortunate Overlook is situated on a high bluff providing dramatic views of Armstead Island and sweeping views of the west and north arms of the reservoir. The site has an aesthetically pleasing stone shelter and walkway and has several interpretive signs describing the Lewis and Clark Expedition and the significance of the events that took place leading to the success of the Corps of Discovery. Other than some additional maintenance items including replacement of the roof on the stone shelter, the site appears to need few improvements. The facility is experiencing steadily increasing visitor use due to the increased attention associated with the bicentennial celebration of the Lewis and Clark Expedition.

<u>Horse Prairie Campground</u> - The Horse Prairie Campground has a west and east camping loop divided by a small gully. The west unit has a boat ramp leading into a small cove that is usable during high reservoir levels. There are several impromptu roads leading off the main access roads that have been used increasing by ORV users. A jack leg fence has been erected along portions of the access road perimeter to help protect the surrounding area from loss vegetative cover due to random ORV use, however the fence has not fully blocked access to several areas that continue to be damaged by ORV users or other vehicles.

The restroom facilities are ADA accessible however the pathways leading to them and most campsites are not fully accessible.

<u>Cameahwait Campground</u> - Cameahwait Campground, sometimes called East Cameahwait to distinguish it from the West Cameahwait Campground, provides several camping sites with small wind shelters. The facilities show signs of age and deferred maintenance particularly the older wind shelters. Most of the facilities are not universally accessible and the site is one of the first to loose direct water front when reservoir levels recede in summer because it is located on the shallow upper end of the reservoir.

<u>West Cameahwait Campground</u> - The West Cameahwait Campground is a campground primarily oriented toward small groups because of its small size and because the site has an existing shelter. The site is adversely affected by lower lake levels since it is on the uppermost portion of the reservoir near the mouth Horse Prairie Creek. However the site is situated along a low-lying, somewhat protected, open field that stays fairly green late into summer. This location is used every August for a historic reenactment of the meeting of Lewis and Clark Expedition and the Lemhi Shoshone Indians which occurred at Camp Fortunate.

<u>Hap Hawkins Campground</u> - The Hap Hawkins facility is primarily a group camping facility that has a large community shelter for both cooking and eating. The facility is the most remote of all sites on the reservoir and offers a large degree of solitude. It also has a somewhat degraded outdoor amphitheater facility but otherwise most of the facilities are in good condition. The toilet facility is ADA compliant however the pathways to the toilets are not currently accessible.

<u>Lone Tree Campground</u> - This campground facility is a popular site used by a wide array of overnight users, many of who come in groups from Idaho. Thus the site is also commonly known as "Little Idaho". The facility is flat and grassy with wind shelters available at most sites. An informal boat launching area is available for car top boat launching or other small craft during higher reservoir levels.

<u>South Shore Fishing Access</u> - This site consists solely of a signed access road and large and flat shoreline area that provides access to anglers during higher water levels. Currently there is no formal parking area however the site is very lightly used.

<u>Red Rock Fishing Access</u> - This site consists of gravel parking area and several informal trails leading to the lower Red Rock River and upper reservoir shoreline, mostly accessible during periods of high reservoir levels. An old railroad and roadbed provide access to the reservoir shoreline. The site is also used by Montana Fish Wildlife and Parks to access their fish trapping facility on lower Red Rock River. The abundance of riparian and wetland vegetation provide good habitat for some wildlife species and the site is also used by wildlife enthusiast such as bird watchers.



<u>Armstead Island Campground</u> - This site is a boat-in only campground and day use area on the west side of Armstead Island. Boats must be able to anchor or pull up onto the beach since no mooring buoy is presently available. Most of the facility components are in disrepair due to age and a lack of maintenance. Because there is no potable water source and because the toilet facility is in bad shape, the site has somewhat limited appeal.

2.7 Transportation Resources

Access to Clark Canyon Reservoir and Barretts Diversion Dam is readily available to motorists from eastern Idaho and southwestern Montana via I-15. State Route 324 provides access from I-15 to the

north and western areas of the reservoir, from which unpaved roads provide access to individual dayuse and campground areas.

2.7.1 Interstate 15 (I-15)



Interstate 15 is a four-lane, divided freeway that runs north/south through the project area and is the major north/south roadway in western Montana and eastern Idaho. I-15 provides for both through-traffic and local traffic movement within the region. The freeway is a limitedaccess roadway, with ingress and egress occurring only at controlled interchanges. Through travel is maintained yearround with snow clearing performed as necessary in the wintertime.

Exit 44 from the freeway is located immediately east of Clark Canyon Dam. This exit is the interchange with SR 324 which runs west from this location, crossing the dam, and continuing along the northern side of the reservoir, then continuing out of the study area to points west and south. This interchange also provides access to a private campground to the east of I-15. From this location, a gravel road proceeds north for a short distance, and then turns west, passing under the freeway, and providing access to Buffalo Bridge fishing access on the Beaverhead River.

Exit 37 from I-15 is located south of Clark Canyon Reservoir and provides access to County Road 188, "Horse Prairie Cut-across Road", (CR-188) which runs northwest from this location.

Exit 56, is located approximately one mile north of Barretts Diversion Dam, and provides access to a frontage road that crosses I-15 and the Union Pacific railroad tracks, to a gravel road that runs south to Barretts Diversion Dam.

2.7.2 State Route 324

State Route 324 is a two-lane, asphalt-surfaced roadway that interconnects with I-15 at Exit 44. From here, SR 324 crosses Clark Canyon Dam, and continues westward through the study area. This roadway provides the most direct route of travel from I-15 to recreation facilities along the northern side of Clark Canyon Reservoir, including: Beaverhead Campground, Beaverhead, Lewis and Clark Day Use, Camp Fortunate Overlook, Horse Prairie Campground, Cameahwait Campground, and West Cameahwait Campground. Access to each of these recreation facilities from SR 324 is via unpaved access roads that are generally in moderate condition and passable to most passenger vehicles.

2.7.3 County Road 188 – Horse Prairie Cut-across Road

County Road 188 (CR 188) is a gravel surface, maintained road that runs parallel to the southwest portion of Clark Canyon Reservoir. The road interconnects with SR 324 approximately three miles west of the reservoir and with I-15 near Red Rock, approximately four miles south of the reservoir.

The roadway provides access from I-15 or SR 324 to Hap Hawkins and Lone Tree Campgrounds, and South Shore and Red Rocks fishing access sites.

2.7.4 Recreation Facility Access Roads

As discussed above, access to individual recreation facilities at Clark Canyon Reservoir is generally via I-15 and either SR 324 and/or CR 188, from which individual access roads provide routes to each facility. Due to the reservoir shoreline contour, facilities along the northern shoreline of the reservoir (i.e., Beaverhead Campground, Beaverhead marina, Lewis and Clark Day Use, Camp Fortunate Overlook, Horse Prairie Campground, Cameahwait Campground, and West Cameahwait Campground) have generally shorter and steeper access roads, while facilities along the southwest shoreline (i.e., Hap Hawkins and Lone Tree Campgrounds, and South



Shore and Red Rocks fishing access sites) have longer, but more gradually sloping access roads. All access roads are unpaved but are maintained in a passable condition for most types of passenger vehicles.

There is an access road to the fishing access location on the west side of Beaverhead River just below the dam. This access road begins at SR 324 west of the dam, and winds down a steep grade to the fishing access near the base of the dam. The road is in moderate condition; however, it is subject to storm water erosion. It contains sharp turns with steeply sloping sides and no guardrails or other safety improvements.

2.7.5 Planned Improvements

Construction of new and upgraded access roads is currently underway for the Camp Fortunate Overlook and west Cameahwait areas. The current Camp Fortunate Overlook access road intersects with SR 324 at a bend in the highway and has limited visibility, which creates safety concerns. The new access road will utilize the marina access road intersection with SR 324 and will run south to the overlook, parallel to and approximately 100 feet east of, SR 324. It is anticipated that Construction of these improvements will be completed in the summer of 2004.

2.7.6 Access to East Side Lands

There is a continuous public access from the Northern most portions of Reclamation's property (N $\frac{1}{2}$ of Sec.5 T.9S. R.10W.), near the Buffalo Lodge to approximately 5 miles south to the Southern most boundary (center of Sec.28 T.9S. R10W.). Reclamation either holds a 60-foot public access right-of-way or owns the 60-foot strip of land where the road is located. There are portions of this area where the road is undeveloped or is a two-track road.

In 1985 and 1986, the General Services Administration sold parcels of Federal land near Clark Canyon Reservoir. These parcels were owned and administered by Reclamation and located east of

Interstate 15. On all parcels of land sold, a public access right-of-way was accepted and reserved unto the United States, its heirs and assigns. This public access right-of-way extends 60 feet Easterly from and paralleling the entire length of the West boundary of the sold parcels. This right-of-way was reserved to the United States, its heirs and assigns on the parcel of land located in the NE1/4SE1/4 Sec. 8 T.9S.R.10W. This public access right-of-way extends 60 feet easterly from and paralleling the entire length of the West boundary of the parcel of lend located in the name paralleling the entire length of the South boundary of the parcel. This right-of-way was secured for public access and access to the radio repeaters.

2.7.7 Other Area Roads

In addition to recreation facility access roads, a small number of undesignated four-wheel-drive roads or trails are located within the Clark Canyon Reservoir area. Unauthorized travel on such roads is prohibited; however, illegal recreational use of these roads occurs on a limited basis and prohibited use is difficult to enforce. Limited signage and fencing has been installed in some areas to educate motorists and deter illegal use of these roads.

2.7.8 Airports

Four airfields located in Dillon, Dell, Wisdom, and Wise River serves Beaverhead County. The Dillon, Wisdom, and Wise River facilities are owned by Beaverhead County, while the Dell Airport is owned by the State of Montana. The largest airport serving the region is the Bert Mooney Airport, which is located in Butte, approximately 70 miles north of Clark Canyon Reservoir. The Bert Mooney Airport serves several incoming and outgoing flights each day, and is a hub for interconnecting flights between smaller local airports and larger regional airports. The major airport in the county is the Beaverhead County Airport located 3 miles north of Dillon. There are no airport facilities on Clark Canyon Reservoir lands.

2.7.9 Railroads

Union Pacific Railroad runs parallel to and east of I-15 through the project area. The railroad travels north from Idaho, and terminates near Butte at an interconnection with the Montana Western Railway. Passenger service has not been identified as available along this route.



2.8 Socioeconomic Conditions

This section discusses the existing economic and social setting of the region within which Clark Canyon Reservoir and Barretts Diversion Dam are located. Much of the information contained herein was compiled by Reclamation in a Recreation Management Conditions Assessment (Reclamation, 2002) completed in August of 2002.

2.8.1 Regional Setting

Clark Canyon Reservoir serves as a regional recreation resource for southwest Montana and northeast Idaho. Day-use visitors are most likely to come from locations within one to two hours drive time of the reservoir, whereas overnight visitors are likely to come from locations of up to 8 to 10 hours drive time away. Estimated driving distances and drive times for selected cities in the region are shown in Table 2.8-1.

Table 2.8-1Estimated Driving Distances and Driving Time to ClarkCanyon Reservoir from Selected Cities					
City	Distance (miles)	Driving Time (hours: minutes)			
Billings, MT	276	4:00			
Bozeman, MT	168	2:33			
Butte, MT	84	1:20			
Dillon, MT	19	:20			
Helena, MT	151	2:20			
Missoula, MT	192	2:50			
Idaho Falls, ID	128	2:00			
Pocatello, ID	175	2:40			
Source: Map Quest					

2.8.2 Population

Beaverhead County is located in southwest Montana, and shares its southern and western border with the state of Idaho. The county covers approximately 5,543 square miles. The county is sparsely populated, with an average population density of less than two people per square mile. In 1990 the population was 8,424 and by 2000 the population had grown to 9,202. In 2000, Beaverhead County contained about one percent of Montana's total population of 902,195.

The town of Dillon, located approximately 19 miles north of Clark Canyon Reservoir, is the county seat, and in 2000 had a population of 3,752. Dillon is the site of University of Montana Western, with a student population of approximately 1,100 during the school year.

Montana and Idaho are geographically large states with relatively low populations. The year 2000 populations of selected locations within or near Beaverhead County are provided in Table 2.8-2.

Table 2.8-2 Populations of Selected Counties and Cities					
County or City	Population (year 2000)				
Montana Counties					
Beaverhead	9,202				
Broadwater	4,385				
Deer Lodge	9,417				
Gallatin	67,831				
Granite	2,830				
Jefferson	10,049				
Lewis and Clark	55,716				
Madison	6,851				
Missoula	95,802				
Powell	7,180				
Ravalli	36,070				
Silver Bow	34,606				
Montana Cities	Montana Cities				
Billings	89,847				
Bozeman	27,509				
Butte	33,892				
Helena	25,780				
Missoula	57,053				
Idaho Counties					
Bingham	41,735				
Bonneville	82,522				
Butte	2,899				
Clark	1,022				
Fremont	11,819				
Jefferson	19,155				
Lemhi	7,806				
Madison	27,467				
Teton	5,999				
Idaho Cities	•				
Idaho Falls	50,730				
Pocatello	51,466				
Source: U.S. Census Bureau 2001a					

2.8.3 Income

In 1999, Beaverhead County had a per capita personal income (PCPI) of \$20,493, and the county ranked 20th out of 56 counties in Montana. The PCPI was 95 percent of the Montana state average of \$21,997, however, only 73 percent of the national average of \$28,546. Montana's 1999 PCPI was ranked 48th in the United States (Bureau of Economic Analysis).

2.8.4 Major Industries by Earnings

In 1999, Beaverhead County residents earned a total of \$112,609,000. This amount was 0.89 percent of the total for the entire state of Montana. The largest industrial sectors of Beaverhead County were government, services, farming, and retail trade. Services, government, and retail trade were the largest segments of the state's economy when measure by earnings. Table 2.8-3 lists earnings by industry for Beaverhead County and the state of Montana.

Table 2.8-3						
Earnings by Industry for 1999						
Sector	Beaverh	ead County	State of I	Montana		
	Amount of Earnings	Percent of Total	Amount of Earnings	Percent of Total		
Farm	\$12,205,000	10.8	\$346,696,000	2.7		
Agricultural, Forestry, and Fishing	\$1,109,000	1.0	\$104,800,000	0.8		
Mining	[confidential]	[confidential]	\$287,003,000	2.3		
Construction	\$7,122,000	6.3	\$952,317,000	7.5		
Manufacturing	\$2,105,000	1.9	\$920,290,000	7.3		
Transportation and Public Utilities	[confidential]	[confidential]	\$1,009,496,000	8.0		
Wholesale Trade	\$2,695,000	2.4	\$633,187,000	5.0		
Retail Trade	\$11,978,000	10.6	\$1,492,441,000	11.8		
Finance, Insurance, and Real Estate	\$9,712,000	8.6	\$744,002,000	5.9		
Services	\$22,057,000	19.6	\$3,413,936,000	27.0		
Government	\$29,740,000	26.4	\$2,728,937,000	21.6		
Total Earnings	\$112,609,000	100.0	\$12,633,105,000	100.0		
Source: U.S. Census Bureau, 2001b						

2.8.5 Major Industries by Employment

Total employment in Beaverhead County accounted for 0.95 percent of the total employment in the state of Montana in 1999. The largest sectors were services, government, retail trade, and farming. These industries accounted for almost 80 percent of all jobs in the county. Table 2.8-4 lists 1999 employment by sector for Beaverhead County and the state of Montana.

Table 2.8-4 Employment by Industry for 1999					
Sector	Beaverhead	d County	State of	Montana	
	Number Employed	Percent of Total	Number Employed	Percent of Total	
Farm	731	14.0	32,122	5.8	
Agricultural, Forestry, and Fishing	132	2.5	8,554	1.5	
Mining	[confidential]	[confidential]	6,498	1.2	
Construction	297	5.7	34,527	6.3	
Manufacturing		2.3	29,287	5.3	
Transportation and Public Utilities	[confidential]	[confidential]	27,327	4.9	
Wholesale Trade	156	3.0	20,784	3.8	
Retail Trade	1,003	19.2	104,951	19.0	
Finance, Insurance, and Real Estate	372	7.1	36,927	6.7	
Services	1,409	26.9	167,868	30.4	
Government	1,011	19.3	83,431	15.1	
Total Employment	5,232	100.0	552,276	100.0	
Source: Bureau of Econo	mic Analysis	•	•	•	

2.8.6 Unemployment

The unemployment situation in Beaverhead County has been better than that of the state as a whole. Table 2.8-5 lists unemployment rates for Beaverhead County, the state of Montana, and the U.S.

Table 2.8-5 Unemployment Rates					
Area 1990 1995 2000					
Beaverhead County	4.8%	4.3%	4.0%		
Montana	6.0%	5.9%	4.9%		
United States 5.6% 5.6% 4.0%					
Source: Bureau of Econo	omic Anal	ysis			

From 1989 through 1997 the poverty rate for Beaverhead County was consistently higher than the rates for Montana and the U.S. Table 2.8-6 lists poverty rates for Beaverhead County, Montana, and the U.S. The poverty rate for Beaverhead County declined by one percentage point in 1997. However, this rate remained more than one percentage point higher than Montana's rate and more than three points higher than the national average.

Table 2.8-6					
		Poverty Rat	es		
Area	1989	1993	1995	1997	1999
Beaverhead County	18.6%	16.8%	17.7%	16.7%	[na]
Montana	16.1%	15.2%	15.8%	15.5%	15.6%
United States	12.8%	15.1%	13.8%	13.3%	11.8%
na = not available					
Source: U.S. Census					

Table 2.8-7							
Poverty Levels							
(household o	f 2 persons und	ler the age of 6	5 with no childr	en under 18)			
Year	Year 1989 1993 1995 1997 1999						
Income	\$8,343	\$9,728	\$10,259	\$10,805	\$11,214		
Source: U.S. Census							
*Household poverty levels are determined annually by the U.S. Department of Health and Human Services (HHS).							
Poverty rates in this table are those published by the U.S. Census based on HHS poverty levels and by							
consideration of household incor	ne, age , family u	init size, and nun	nber of children	under 18.			

In association with the Recreation Management Conditions Assessment (Reclamation, 2002), a computer-based modeling software program was used to present baseline economic conditions and calculate estimated contributions of Clark Canyon Reservoir to the local and regional economy. The IMPLAN² input-output model, using 1996 data for Montana, identified the baseline economic conditions of total industrial output, employment, and labor income for Beaverhead County (see Table 2.8-8). The Beaverhead County output (in 1996 dollars) was \$345,335,000, which amounted to approximately 1 percent of the total output for the state of Montana (\$33,971,919,000). Only 5,218 of the state's 523,325 jobs were located in Beaverhead County in 1996. Employee income for the county was approximately \$95,095,000; less than one percent of the state total of \$9,953,882,000.

2.8.8 Contribution of Recreation at Clark Canyon Reservoir to the Local Economy

In order to estimate the contribution recreation at Clark Canyon Reservoir adds to the Beaverhead County economy, data on visitor use and recreation-related expenditures are necessary.

The five-year average for visitation approximates 59,000 recreation visits annually. However, only those visitors who live outside Beaverhead County and purchase goods and services in Beaverhead

² IMPLAN is the Impact Analysis for Planning model software originally developed for the U.S. Forest Service and maintained and updated by Minnesota Implan Group, Inc. (MIG).

County contribute additional dollars to the county's economy. This is based on the assumption that if recreation opportunities were not available at Clark Canyon Reservoir residents would spend the same amount of money on other recreational activities within (or possibly outside of) Beaverhead County. To account for the regional verses out-of-region visitor use, the annual visitor use figure is reduced to 50,000 recreation visits per year.³



Dillon Railroad Station

Specific data on expenditures by visitors to Clark Canyon Reservoir are not available. Therefore, a benefits transfer procedure provided expenditure data for this analysis. Such data was collected and developed for Canyon Ferry Reservoir, which is located in Broadwater and Lewis and Clark counties. Canvon Ferry is a larger reservoir, but offers similar opportunities for boating, camping and fishing. Both reservoirs are in Montana and share overlapping market areas (i.e., they both draw visitors from some of the same counties in Montana). For the purposes of this analysis, the Canyon Ferry expenditure patterns and data are assumed to be similar and representative of the out-of-region Clark Canyon visitor expenditures.

In addition, the Canyon Ferry expenditure data were collected and developed in such a manner that expenditures by economic sectors were identified. This fact allows the use of this data with IMPLAN to estimate the financial contribution that recreation at Clark Canyon Reservoir adds to the economy of Beaverhead County (i.e., approximately \$252,500 in direct expenditures due to approximately 50,000 recreation visits originating from outside the region).

Total output related to recreation use at Clark Canyon Reservoir was approximately \$335,000 in 1996 dollars. This figure includes direct impacts⁴, indirect impacts⁵, and induced impacts⁶. Recreation supports approximately nine jobs and provides \$142,500 in labor income, as listed in Table 2.8-8. This contribution is less than one-fifth of one percent of the total in Beaverhead County.

³ p. 71 of the Recreation Management Condition Assessment, Clark Canyon Reservoir and Barretts Diversion Dam. Bureau of Reclamation, Montana Area Office, August 2002.

⁴ Direct impacts are the recreation-related out-of-pocket expenditures by visitors from outside the region that cause the change in output, employment, and income in the region.

 ⁵ Indirect impacts are the changes in output, employment, and income caused by industries/firms purchasing goods and services from other industries within the region.
 ⁶ Induced impacts are the changes in output, employment, and income caused by new household income

⁶ Induced impacts are the changes in output, employment, and income caused by new household income generated by the direct and indirect expenditures in the region.

Table 2.8-8 Recreation Contribution to the Local Economy							
Economic MeasureClark CanyonBeaverheadClark CanyonReservoirCountyReservoir Percent of							
Industry Output	\$335,456	\$345,335,000	County 0.10%				
Employment	9.3 jobs	5,218 jobs	0.18%				
Labor	\$142,569	\$95,095,000	0.15%				
Source: IMPLAN and Bu	Source: IMPLAN and Bureau of Reclamation						

2.8.9 Net Benefits of Recreation

Data limitations require certain generalizations and assumptions to estimate the net benefits of recreation⁷ for Clark Canyon Reservoir. First, since fishing is the most popular activity of visitors, it is assumed that all visitors participate in this activity for at least some part of their visit to the reservoir and that the value for fishing will adequately represent the value for total recreation use by individuals at the reservoir. (A visitor to Clark Canyon Reservoir may participate in several different activities during the course of their visit, however, fishing has been selected as the proxy for all activities for the purposes of simplifying the analysis and assumptions.) The second assumption is that the most recent five-year average of visitation and use represents a reasonable expectation of future annual use (i.e., 59,000 recreational visits per year).

Benefit transfer research identified the average net benefits for fishing within the intermountain region, which includes Montana, as being \$40.82 per person per day in 1996 dollars (Rosenberger and Loomis, 2000). Accounting for inflation (using the Consumer Price Index from the Bureau of Labor Statistics) brings this value up to approximately \$46 in 2001 dollars. Therefore, the estimated annual value of fishing within intermountain region is approximately \$2,714,000 in 2001 dollars.

2.9 Public Services and Utilities

Clark Canyon Reservoir and Barretts Diversion Dam are located in Beaverhead County, Montana. Public services within the area include those offered by public agencies as well as private companies. The following sections discuss a number of public services provided by or provided for areas within Beaverhead County.

2.9.1 Beaverhead County Police Protection

The Beaverhead County Sheriff's Office currently employs seven sworn officers, two reserve officers, four full-time 911 dispatchers, and other personnel that are funded through the general revenue fund for the County. The ratio of sworn officers to population is 1:1,286.

<u>State and Federal Agencies</u> - In addition to local enforcement agencies, there are a number of State and Federal agencies that provide services and coordinate with the Beaverhead County Sheriff's Office. The Montana Highway Patrol provides traffic enforcement for all state highways (primarily

⁷ Net benefits or value of recreation equals the amount of money an individual would be willing to pay over and above the amount they actually do pay to participate in a recreational activity. Economists also refer to this amount as consumer surplus.
I-15 in Beaverhead County) and county roads MFWP, the United States Forest Service and the BLM provide wardens and law enforcement officers to enforce regulations on non Reclamation public land (Beaverhead County, 2003)

<u>Beaverhead County Search and Rescue</u> – The County Search and Rescue is a 60 member volunteer organization that operates under the direction of the Sheriff's office.

According to the County, as the needs for service in all areas of operation increase, so will the need for additional staffing. The facilities currently being utilized will need to be enhanced in the near future. Within the next 10 years, the need for larger administrative facilities will have to be addressed. As jail regulations become more stringent or mandatory, the present detention center will most likely be out of compliance. Other areas of concern will be additional jail staff and patrol staff to meet the needs of the public. (Beaverhead County, 2003)

2.9.2 Emergency Medical Services

Beaverhead Emergency Services, Inc. (Beaverhead EMS) provides emergency response to the City of Dillon and the surrounding area. Beaverhead EMS consists of 30 volunteers, which include 22 Emergency Medical Technicians (EMT's), five EMT Intermediates, one EMT Paramedic, and Registered Nurses. Beaverhead EMS has three advance life support level ambulances and one extrication truck, and average 300 to 400 runs per year, including transfers from Beaverhead County to hospitals in Missoula, Butte, and other regional medical facilities. (Beaverhead County, 2003)



Downtown Dillon

The Wisdom Volunteer Fire Department also operates an ambulance, averaging 11 to 12 runs per year over the past 5 years. The Wise River Volunteer Fire Company provides full ambulance services with five EMT's averaging 15 to 16 runs per year during the period between 1995 and 1999. The Grant Quick Response Unit provides emergency response and non-transport emergency medical services to the Grant/Horse Prairie area. The Lima Volunteer Fire Department provides ambulance service in the southern portion of Beaverhead County. Air ambulance and life flight services are available out of Idaho Falls and Missoula. (Beaverhead County, 2003)

2.9.3 Medical Facilities

The nearest medical facility to Clark Canyon Reservoir and Barretts Diversion Dam is Barrett Memorial Hospital, which is a 31-bed medical facility located in Dillon. Medical services available include intensive care, surgery, in-patient and out-patient care, physical therapy, laboratory services, x-ray/ultrasound/CT/MRI, emergency services, and other related services. (Beaverhead County, 2003)

2.9.4 Volunteer Fire Districts and Companies

Three organized fire districts and two fire companies provide fire protection throughout the County. Volunteers man all of these organizations. The districts are largely funded with property tax revenue, although all of the departments and fire companies rely heavily on donations, fund raisers, grants, and support from the Natural Resources and Conservation (NRCS) which, in most cases, is responsible for wild land fire protection. While rural fire districts or fire companies cover most of the county, there are portions of the county that do not have adequate protection for private property. In these areas, the County Fire Warden (Beaverhead County Sheriff) has jurisdiction and authority. (Beaverhead County, 2003) Reclamation is a participant in a multi-agency fire agreement with Federal, State, and County agencies that was established in 1982 and modified in 1994. The objectives of this agreement are coordination, prevention, detection, dispatch, and suppression of fire between the agencies.

The Beaverhead County Wild land Fire Management Plan, updated in 1999, provides for cooperative and mutual aid agreements between all of the volunteer departments and companies in the county and the DNRC and USFS. The Beaverhead County Disaster Plan allows for mutual aid between all of the fire departments and companies for non-wild land situations.

A number of challenges make residential fire fighting difficult for the VFD, particularly in the wild land residential interface. Lack of defensible space, limited water supplies and building materials that are not fire resistant make it difficult and dangerous for volunteers to adequately provide protection in these areas.

<u>Fire District #1</u> - The Lima Volunteer Fire Department provides fire protection to the southern end of the county roughly from Clark Canyon Reservoir south to the Idaho border.

<u>Fire District #2</u> - The Dillon Volunteer Fire Department provides fire protection to Dillon and the surrounding Beaverhead Valley. The Grant Volunteer Fire Department services the Horse Prairie area south and west of the portion of Clark Canyon Reservoir which is in District #2.

<u>Fire District #3</u> - The Big Hole area and portions of the Grasshopper Valley are covered by this fire district. The Wisdom Volunteer Fire Department, Jackson Volunteer Fire Department, and Grasshopper Valley Volunteer Fire Company all serve portions of this district. The Grasshopper Valley Volunteer Fire Company was recently formed in 1999 and has constructed a fire hall, and also will service portions of the Grasshopper Valley that are currently not in any fire district.

<u>Wise River Volunteer Fire Company</u> - The Wise River Volunteer Fire Company provides fire protection to the northern part of the county along the Big Hole River and Highway 43 from Fish Trap to Divide.

All of the departments have relatively new buildings. Equipment upgrade and replacement is needed by all departments, but it is almost cost prohibitive for some of the smaller departments. (Beaverhead County, 2003)

2.9.5 Solid Waste

The Beaverhead County Solid Waste District operates the Beaverhead County Landfill located five miles west of Dillon. The landfill is licensed for municipal waste and provides service to Beaverhead and Madison Counties, and parts of Silver Bow and Deer Lodge Counties.

The existing landfill has capacity until the year 2063 based on current volumes of waste of approximately 11,000 tons per year, and the landfill has the capability for future expansion. (Beaverhead County, 2003)

2.9.6 Electric Utilities

Two electric utility companies serve Beaverhead County. The Montana Power Company serves an area of 97,540 square miles, including areas within Beaverhead County, with a total of over 280,000 customers. The Montana Power Company has a total capacity of approximately 100,000 kilowatts (kW) with a peak demand of approximately 15,000 kW. The second electric utility that serves portions of the county is the Vigilante Electric Co-op. Vigilante Electric has a system capacity of approximately 30,000 kW, and serves rural areas including Melrose, Glenn, Divide, Wise River and Wisdom. (Beaverhead County, 2003)

Clark Canyon Draft Resource Management Plan for Clark Canyon Reservoir and Barretts Diversion Dam

Chapter 3 Needs, Constraints, and Opportunities

3.0 Needs, Constraints, and Opportunities

During the RMP scoping process, various opportunities and constraints associated with the management of resources and recreation at Clark Canyon Reservoir and Barretts Diversion Dam were identified. These needs, opportunities, and constraints were evaluated and considered as part of development of the project alternatives. The types of opportunities and constraints considered include financial, legal, reservoir carrying capacity, and environmental conditions (topography, cultural resources, water quality, endangered species).

3.1 Recreation Resources

Needs

Use of Clark Canyon Reservoir recreation resources is limited by several factors, including a low regional population base, limited and aging facilities, little to no shade or shelter, and fluctuations in reservoir levels, which are based on current hydrologic cycles, contractual requirements, and down stream environmental needs.



Constraints

Constraints identified in the conditions assessment report include operation of the reservoir for flood control, irrigation, and other downstream purposes, which limits Reclamation's ability to manage reservoir storage and releases for the purpose of enhancing recreation. As identified in the report, Reclamation has limited flexibility to change the historic operation of its water projects because of its existing contractual obligations. In addition, Reclamation lacks specific recreation authority to manage specifically for recreation. Through P.L. 89-72 Reclamation can develop new recreation resources in cooperation with non-federal government entities; however, if a managing partner is not obtained, Reclamation is restricted to providing only minimum and basic recreation facilities.

Opportunities

Recreational opportunities abound at Clark Canyon Reservoir. The reservoir is located roughly halfway between the cities of Butte, MT and Idaho Falls, ID, and is immediately accessible from Interstate 15. Clark Canyon Reservoir has one of the most productive reservoir fisheries in the State of Montana, and with an abundance of open space and easy access.

The August 2002, "Recreation Management Condition Assessment" for Clark Canyon Reservoir and Barretts Diversion Dam identified several opportunities. One of the recreation opportunities identified was cost sharing for recreation purposes with other federal, state and local agencies could increase the capability of Reclamation to successfully manage Clark Canyon Reservoir and Barretts Diversion Dam. Also identified were partnerships with private organizations to better maintain the recreational amenities available at the reservoir.

3.2 Natural Resources

Needs

Sport fishing could benefit from improved water quality through shoreline stabilization. In addition, wildlife and natural habitats could also benefit from closing ORV trails, establishing better-defined parking and camping areas, and improving vegetative cover to reduce noxious weeds. Improving habitat could also increase the number and types of birds and other animals using the reservoir shoreline and riparian areas, which in turn could likewise increase interest in the designated wildlife viewing area previously discussed.

Constraints

Clark Canyon Reservoir is operated for irrigation water supply and flood control under existing authorities. Under these authorities the runoff from upstream is primarily stored or released as needed to meet irrigation demands. The operation of the reservoir is not a subject of this RMP. As a result, the changing water level is a major constraint to provide potential fish and wildlife enhancement measures. The extensive on going efforts to control the spread of noxious weeds, also constrains the ability of Reclamation to focus on other enhancements. Constraints that limit or provide challenges to natural resource management include steep slopes on portions of the shoreline, shoreline erosion areas, noxious weed infestation, and potential conflicts between wildlife and recreation users. In addition, the carrying capacity of the area to support wildlife also limits the amount and type of wildlife present.

Opportunities

There are numerous opportunities to improve and protect the existing natural resources at Clark Canyon Reservoir and Barretts Diversion Dam. The existing designated wildlife viewing area located on the Beaverhead River just below the dam could be enhanced for the public to enjoy with relatively minor access improvements. In doing so, this enhancement could help add to the local economy by increasing visitation to the area of visitors coming to the wildlife area for reasons other than fishing.



Federal laws require that Federal lands be managed for protection of cultural resources and enhancement of species listed under the Endangered Species Act. At Clark Canyon Reservoir, protection from bald eagle disturbance, particularly during the foraging seasons, may limit some recreation activities. In addition there are several candidate species on nearby lands and any future actions that would adversely affect these species will not be allowed. All actions near eagle foraging areas will need to consider potential effects to ESA and Federal Candidate species.

3.3 Lands

Needs

There is a need to maintain public access to reservoir lands and surrounding public lands. This includes all Reclamation lands owned and easements located East of I-15. A need exists to continue

the management of existing lands for fishing, hunting, and wildlife viewing. Provide administrative access and use of MFWP Fish Trap facility. Near Clark Canyon Dam, there are continuing needs to pave the main access road, rehabilitate drainage and erosion features, and add guardrails on road to Beaverhead River Fishing access below dam. There is a need to increase noxious weed control along old railroad grade and south boundary near Red Rock Creek, Horse Prairie open lands, and around all recreation sites. For the fences on the South Shore Lands, there is a need to repair, rebuild and extend sections including creating more adaptable passage for Antelope. On all lands there are needs to: 1) Exclude ORV trails and vehicles in sensitive areas. 2) Limit firearm usage to shotgun, handgun, and black powder. On the lands around Horse Prairie Creek Mouth there is a need to develop a grazing plan to reduce residue built up and stimulate new vegetative grow through prescribe grazing by livestock. Finally there is a need to plant willows and cottonwoods in the riparian areas of Horse Prairie Creek lands.

Constraints

Reclamation has limited budget and personnel resources to construct and maintain improvements, and enforce regulations. Reclamation is also limited in its ability to apply effective chemicals and or biological control to noxious weeds in a recreation setting. In the effort to preserve and encourage vegetation the competition from noxious weeds, other vegetation, and lack of moisture limits success in establishing or preserving important native vegetation.

Opportunities

There are numerous opportunities to partner with dedicated hunting clubs; sports based groups, wildlife clubs, and youth groups such as, Boy Scouts, to obtain volunteers in fence repair, tree and willow planting. Opportunities exist to enter into cooperative agreements with adjacent landowner(s) to lease grazing rights. Through more public awareness campaigns and partnerships with recreationists and nature lovers, reporting of abuse from ORVs will result in less ORV abuse and the spread of noxious weeds. Through partnerships and coordination with County, State, and Federal agencies scheduling of road upgrades can reduce cost and increase opportunities for improving access roads.

3.4 Cultural Resources

Needs

While there have been some cultural resource surveys at several locations around Clark Canyon Reservoir, there is a need to perform further surveys to identify and protect other resources not yet identified.

Constraints

The primary constraint to effectively manage the cultural resources is related in the low funding levels. Additional comprehensive evaluations have been provided in the past to better guide cultural resource management.

Opportunities

Clark Canyon Reservoir lands contain both historic and prehistoric resources of importance. These resources provide great opportunities for interpretive displays and programs that can showcase both Native American cultures and historic settlements and past resource uses of the area.

Research conducted in the area has revealed prehistoric sites that demonstrate extensive Native American utilization of the region. Artifact scatters and quarry sites show the area was used for stone tool production, bison kills, and butchering sites demonstrate meat procurement, and camps with fire hearths and grinding stones demonstrate food preparation. Although none of the prehistoric sites at Clark Canyon Reservoir have been nominated to the National Register of Historic Places some may be eligible since they may be likely to yield information important in prehistory or history.

The historic period at Clark Canyon Reservoir dawned in 1805 when the Lewis and Clark Expedition entered the area. Their meeting with the Lemhi Shoshoni Indians at "Camp Fortunate" at a location now under the waters of Clark Canyon Reservoir was an important factor in the success of the expedition. The Lemhi Shoshoni Indians were not the only tribe known to have utilized the area, research has reported usage by the Bannock, Flathead, Pend Oreille, Nez Perce and Blackfeet as well.



Lewis & Clark Monument at Clark Canyon

Discovery of gold by John White's party in 1862 started the gold rush to Montana. Bannack and Virginia City, located northwest and northeast of Clark Canyon became leading mining towns.

The Gilmore & Pittsburgh Railroad (G & P) was constructed from the mainline of the Utah and Northern up Horse-Prairie Creek to the mines at Salmon and Gilmore, Idaho. A community named Armsted, Montana developed at the junction of the two railroads. Portions of the grade were so steep, that the gained the nickname "Get Out and Push". The (G O & P) was not profitable, and the tracks were removed in the early 1940's. The post office remained open at Armstead until 1962.

Few historic remains are located on Reclamation land adjacent to the Reservoir, and none are on the register of historic places. However, given this rich history, there are several areas on Reclamation lands where interpretation opportunities exist. There are enough landscape features that directly relate to the specific historical and prehistorical documentation of the area. These areas provide prime settings for interpretive displays or view points keyed to an interpretive brochure or booklet. The Federal Government is required by law and regulations to protect and preserve significant cultural resources. To this end, Reclamation's undertakings are subject to compliance required by the National Historic Preservation Act and its regulations found in 36 CFR Part 800. These mandate require that the Government consider the effects of its action on prehistoric and historic resources before implementing those actions.

3.5 Operations, Maintenance, Enforcement, Public Health and Safety

As identified earlier, there are several opportunities for Reclamation to increase and enhance the existing recreational and wildlife resources. However, with Reclamation's limited authority under Public Law 89-72, major improvements of recreation operations and maintenance activities at the

reservoir would have to use cooperative agreements with a managing partner(s). This would allow for collection of user fees to help fund capitol improvements and recreational facilities maintenance. Public health and safety would also be improved with improved facilities.

3.6 Funding Opportunities

The ability to retain recreation-related fees to offset operation and maintenance expenses is an important aspect to consider when managing recreation activities on Federal lands. Currently, Reclamation does not collect user fees at Clark Canyon Reservoir. Reclamations budget is based on annual Congressional appropriations.

P.L 89-72 authorizes Reclamation to enter into recreation management agreements with non-federal entities. This act also authorizes the transfer of management of project lands or facilities to these agencies with terms and conditions that best promote development and operation of the land and facilities for recreation purposes in the public interests (Recreation Management Condition Assessment, 2002). While this presents an opportunity to improve the recreation and natural resources at Clark Canyon Reservoir and Barretts Diversion Dam, a limiting factor is the ability to find a willing and suitable partner. To date, Reclamation has only had limited success in attracting a private concessionaire for operating a marina primarily because of the need for key capitol improvements that would allow for a concessionaire to successfully manage a marina and provide a reasonable investment and return. Also, due to the recent drought and low water levels over the past few years, it is difficult to attract a large amount of recreationists to the marina.

Clark Canyon Draft Resource Management Plan for Clark Canyon Reservoir and Barretts Diversion Dam

Chapter 4 Resource Management Plan Actions and Elements

4.0 Resource Management Plan Actions and Elements

This chapter provides general and specific actions proposed for the management of lands and resources at Clark Canyon Reservoir and Barretts Diversion Dam. General actions are those that are broad based management actions that include overall unit-wide goals and objectives and apply to all Reclamation lands around the Reservoir and at Barretts Diversion Dam. Specific actions are management actions formulated to be site-specific, prescriptive-based actions directed at specific purposes for specific areas. Management actions are specific tasks intended to guide Reclamation staff and managing partners to proper management of Reclamation lands. These actions are intended to be implemented over the next 10 years. Inclusion of these actions does not ensure the funding, staff, or equipment will be available to implement these actions, nor does it obligate Reclamation to implement. The following lists provide the four main categories of actions, the elements and subcategories of these actions are further described in this chapter.

- Recreation (Section 4.2) includes improvements and management considerations for the developed shoreline and dispersed recreation sites;
- Land Resources (Section 4.3) includes wildlife and vegetation management, erosion control, and water quality.
- Cultural Resources (Section 4.4) includes archaeological and paleontologic resources and Traditional Cultural Properties
- Operations, Maintenance, and Enforcement (Section 4.5)

4.1 Plan Formulation

In formulating the management actions determined to be most appropriate for effective and prudent management of the lands and recreation facilities at Clark Canyon Reservoir and Barretts Diversion Dam a screening process was used to analyze a range of possible development, redevelopment plans and natural resource management planning choices for the lands around Clark Canyon Reservoir. This screening process led to a final set of actions presented in this chapter.

The findings of Reclamation's August 2002 "Recreation Management Condition Assessment" provided further information on specific management actions deemed relevant and appropriate to address future development options at Clark Canyon Reservoir. From these analyses, a range of management actions was identified.

The screening process to select possible management actions involved several steps including the consideration of the following factors:

- Opportunities for natural resource management and development,
- Constraints for natural resource management and development,
- Reclamation policies and authorities,
- Existing evaluations such as the 2002 "Recreation Management Condition Assessment,"
- Public input obtained during the fall of 2002 scoping meetings and through the ongoing public involvement program; and
- Agency and local government needs and requirements.

Through a screening process that carefully evaluated each of the above listed factors, a series of alternative management actions at each recreation and/or management unit were identified and documented in the Resource Management Plan Environmental Assessment for Clark Canyon Reservoir and Barretts Diversion Dam (EA). In the EA, the management actions were combined into no-action, moderate resource development, and maximum resource development. The combined actions identified by recreation site and land area at Clark Canyon Reservoir provided a broad range of choices in guiding the management and near term development or redevelopment of recreation facilities within the project area



The No-Action Alternative represents a continuation of pre-RMP management practices. On-going

maintenance and associated replacement or repair of existing facilities would continue under this

alternative. Some replacement of facilities/structures associated with the Lewis & Clark celebrations would likely continue in the short term, however, deferred maintenance would be done when funds became available. This alternative also includes a new access road into the Camp Fortunate overlook from State Route 324 for safety and since funding for this work is already secured and construction is underway.

Alternative A consists of a series of actions labeled the "moderate resource development alternative". This alternative includes improvements and management actions that could reasonably be accomplished within the 10-year planning period based on previous levels of funding. Elements of this alternative seek to solve some of the existing problems with facility layout, access, or deferred maintenance. The various actions under this alternative are further outlined in this chapter.

Alternative B is the "maximum resource development alternative" and includes a series of improvements and management actions that would require substantial upgrading of reservoir recreation facilities, and in many cases, expanding existing facilities and bringing facilities up to current standards. This alternative incorporates many of the same elements identified in Alternative A, but includes additional elements addressing more extensive improvements and needs identified during the scoping process. Alternative B includes the construction of new facilities at several campgrounds and day use areas. Alternative B is described in detail in Section 2.2.3 of the Environmental Assessment.

At the end of the screening process, a single alternative was chosen based on several factors including demonstrated need, extent, type, and intensity of likely environmental and socioeconomic effects. It also included consideration of the ability to secure funding as well as past management history and likely effectiveness of the action(s). The alternative selected is Alternative A and it is analyzed in the EA document as the moderate resource development alternative. This alternative represents the proposed plan as presented in Figure 4.1-1 (see figure at the end of chapter 4). Goals and objectives of the Clark Canyon Reservoir Proposed Plan are in the following sections.

4.2 Recreation Resources Goals and Objectives

This section identifies goals, objectives and where possible actions applicable to guide future recreation resource management of Clark Canyon Reservoir lands. These goals and objectives are intended to complement site-specific land area management prescriptions that follow.

The goals for recreation resource management include provision for safe and suitable road access to designated recreation areas.

For this goal, the objectives and general actions include:

- a. Provide access management and parking within designated public use sites.
- b. Ensure motorized vehicles stay on designated roads and vehicle users are to obey signs including closed area signs. There is to be no vehicular use of the exposed lakebed unless special permission is obtained from Reclamation officials in the Montana Area Office in Billings.
- c. Maintain the existing public access right-of-way that is currently in place. One area of interest is East side lands. (See Figure 4.1-1)
- d. Improve Beaverhead Fishing access road to prevent erosion and for safety concerns.

A second major goal for recreation resource management includes provision for minimum and basic recreation site improvements, consistent with demand, available funding and carrying capacity of the resource base.

For this goal, the objectives and general actions include:

- a. Ensure compliance with existing federal legislation and regulations in accordance with ADA for designated, developed sites and facilities.
- b. Accommodate future increases in recreation demand and visitation by expanding with help from partner, or rehabilitating existing sites as appropriate.
- c. Actively manage recreation services, especially restroom and trash services at popular sites used during holiday weekends.
- d. Continue to upgrade the existing facilities with more maintenance free, uniformly accessible equipment.

Managing litter and providing litter control is also an ongoing maintenance and social-environmental issue. Litter receptacles, including dumpsters, will not be provided at dispersed recreation areas such as the Sacajawea Day Use. Recreationists will be expected to follow a "pack it in- pack it out" concept in carrying out all garbage and litter resulting from their activities. Policy statements on

litter and garbage removal will be included in visitor information materials and programs on land management.

In addition to the general actions for managing recreation resources, there are land management prescriptions specific to the different recreation facilities at Clark Canyon Reservoir and Barretts Diversion Dam. The specific management actions for each recreation site are provided as follows:

4.2.1 Barretts Diversion Dam



Barretts Diversion Dam is a very popular day use area used for activities including weddings, group picnicking, and fishing. Several relatively simple improvements have been identified that would make the area more usable and accessible. Improve the lawn areas near the diversion dam by fertilization. Through assistance from partner construct an ADA accessible path and parking area leading to the existing toilet.

4.2.2 Beaverhead River Fishing Access

Beaverhead River Fishing Access is a frequently used site situated immediately below Clark Canyon Dam. Compared to most other areas around the reservoir, this area is lush with vegetation and is used by anglers, campers, bird watchers, and other wildlife observers. The area is also a popular put-in location for anglers with drift boats. Access to the site is from a steep, winding road that begins near the top of Clark Canyon Dam. The recommended RMP improvement for this area is to improve the access road down to the fishing/camping area to address erosion and safety concerns.

4.2.3 Buffalo Bridge Fishing Access

The Buffalo Bridge Fishing Access is heavily used during the fishing season. The site is not signed, but is well known and accessed from the east side of I-15 by crossing under the freeway bridge that spans the Beaverhead River just below the dam. During the fishing season the area is often congested and difficult to use, especially when anglers are trying to launch their boats. To help alleviate this and other problems, signs would be installed to better delineate the parking areas below the bridge, as well as, in the area above the bridge where cars and trailers typically park. The existing informal paths would be replaced with an accessible walkway to the toilet.

4.2.4 Marina

The existing marina is limited in use by several factors, primary being the reservoir levels are unusually low and there is no current operator (concessionaire). However, the marina facility is still used for RV camping. Until the spring of 2002, the facility was operated by a private operator under a concession agreement with Reclamation. The concessionaire provided gas, oil, boating supplies, mooring slips, and some groceries. In the past 15 years, there have been several different concessionaires who attempted to operate the marina successfully, but they all had difficulties making it a successful business venture. Currently, there appears to be little interest given the

investment required to upgrade the facility to modern standards and the large annual fluctuations in the reservoir level. The drought of the past 5 years (1999-2003) has drastically reduced the inflow and water into the reservoir. Consequently the marina has been closed and the boating use has dropped significantly. The need for an operator has been identified as critical if the marina is to remain an amenity at the reservoir, but Reclamation recognizes that improvements to the current facility will be needed in order to attract a new concessionaire. The RMP improvement for this area addresses some of the needs at the marina facility such as improving the water supply system, providing more shade trees and irrigation, and making some minor improvements to the RV camping and parking sites. The RMP also proposes continuing the use of the area, managed by campground host/volunteers, for overnight camping with nominal fees.

4.2.5 Lewis & Clark Campground

The existing Lewis & Clark Campground site provides for overnight camping uses, but the area receives low use. The RMP calls for eliminating the overnight camping uses as a result of the recent relocation of the Camp Fortunate Overlook access road through the site. Also, Reclamation will install a new sign to designate it as a day use area and the existing units can be used for day use shelters and picnic sites.

4.2.6 Beaverhead Campground

Beaverhead Campground is the most popular and conveniently accessible camping site at the reservoir. The plan is to implement improvements to provide improved low-water boat access, upgrade existing picnic shelters, improve ADA access to the toilets, improve landscaping near existing shelters, and better delineate the low-water boat ramp parking.

4.2.7 Camp Fortunate Overlook



The Camp Fortunate Overlook is increasingly used for educational and interpretative uses celebrating the Lewis and Clark Expeditions of 1804-1806. The facility is also used for passive recreation (limited trail use) and provides excellent views of the reservoir. The plan calls for improvements to the interpretive facilities to better reflect the historic qualities of Camp Fortunate.

4.2.8 Sacajawea Day Use

The Sacajawea Day Use is a dispersed recreation site that currently consists of a series of dirt roads leading from SR 324 to the waters edge. The site is somewhat barren other than a few juniper trees standing along side the footprint of a cabin that once existed in the area. The proposed improvement to the Sacajawea Day Use is to close down non-essential roads within the area and replace a toilet.

4.2.9 Horse Prairie Campground

Horse Prairie Campground is used for camping and fishing access on the north side of the reservoir. The plan calls for improvements to the parking area and camp site designations, install new jack-leg fencing to restrict ORV traffic, and repair the existing boat ramp, as well as, implement other measures to address areas of erosion along the portion of Horse Prairie Creek within the campground.

4.2.10 Cameahwait and West Cameahwait Campground

The RMP improvements for Cameahwait and West Cameahwait Campground include adding new ADA access to toilets and shelters. The non-essential roads will be closed to minimize ORV use.

4.2.11 Hap Hawkins Campground

Improvements for this area are to refurbish the existing group shelter and improve access to toilets by providing accessible walkways from the parking area.

4.2.12 Lone Tree Campground

Lone Tree Campground is a popular group camping area that serves as a base camp for fishing access to the reservoir, as well as, for large gatherings of campers in multiple groups that desire shared facilities. Improvements identified for this site include adding signage to delineate camping areas, improve the water supply, and add an ADA toilet that is accessible from the parking area.



4.2.13 Armstead Island

The isolation and lack of vehicular access to Armstead Island limits the type of improvements that can reasonably be provided. One fundamental and reasonably achievable improvement is to replace existing picnic tables with new tables.

4.2.14 South Shore Fishing Access

No improvements have been identified for the South Shore Fishing Access area as the area currently appears adequate to serve the public given the low level of use typically found at this location.

4.2.15 Red Rock Fishing Access

The Red Rock Fishing Access area is heavily used during peak fishing periods. To better meet peak demand, the parking area would be expanded and an accessible toilet would be provided with a walkway from the newly expanded parking area.

4.3 Land Resources Goals and Objectives

This section identifies goals, objectives and possible actions applicable to guide Reclamation's future natural resource management of project area lands. These goals and objectives are intended to complement site-specific land area management prescriptions that follow.

The goal for land management dealing with surface water runoff is:

Protect and improve water quality and limit shoreline erosion and sedimentation.

For this goal, the objectives and general actions include:

- a. Include erosion control measures (e.g. straw mulches, sediment traps and filter fabric) in the design and construction specifications for any proposed development. Contract specifications will contain measures to prevent erosion and sediment-laden runoff from leaving the work sites.
- b. Discourage and prevent any future grazing practices that allow cattle to enter the reservoir. Control cattle movement around other water bodies to minimize sedimentation and direct bacterial contamination of incoming source waters.

The goal for Fish and Wildlife habitat is:

Protect and enhance important vegetation fish, and wildlife habitat values.

For this goal, the objectives and general actions include:

- a. Protect existing trees and promote the establishment of native trees and shrubs.
- b. Require that any permitted livestock grazing on Reclamation land does not result in a significant adverse effect on existing valuable vegetation and wildlife habitat or the rate of recovery of habitat previously damaged habitat.
- c. Continue program to control noxious weeds and promote the growth of native species.
- d. Identify methods to restore upland vegetation/habitat damaged by vehicular access focusing on areas where access has been terminated.
- e. Identify locations where State-listed noxious weeds, invasive exotics, and other plants are a problem requiring action.
- f. Assist in bald eagle recovery efforts and other bird nesting and foraging studies as well as other wildlife studies that are active in the region over the next 10 years.
- g. Limit firearm usage to shotgun, handgun, and black powder on all Reclamation lands surrounding Clark Canyon Reservoir.

h. Provide Raptor protection on utilities.

In addition, encroachment and trespass on Reclamation lands is not to be allowed. Yearly inspections will be conducted on an informal basis and a formal review will be conducted every 5 years. Agricultural and grazing use of Reclamation lands will be allowed only as a land management tool. Any dumping problems will be followed up with local law enforcement officials if necessary. Boundary identification will be improved where feasible with installation of signs, monuments, and fences.



In addition to the general actions for managing natural resources, there are land management prescriptions specific to the different open land areas surrounding Clark Canyon Reservoir. The specific management actions for these land areas are provided as follows:

4.3.1 Horse Prairie Creek Mouth

The area near the mouth of Horse Prairie Creek as it enters the reservoir has the potential to provide added benefits to recreational users and wildlife that frequent the area. Currently, the ecological condition of the vegetation has declined in productivity and vigor. Weeds such as Canadian thistle and whitetop have increased in density. The sub-irrigated and wet meadow sites have very substantial amounts of litter accumulation that have essentially smothered the perennial plants such as rush and sedge. Most of the willows and cottonwoods in the area are dead or nearly dead. Even the wild rose is struggling to survive. One of the most straightforward measures that are often found to be successful in similar settings is to plant willows and cottonwood along waterways to help maintain plant diversity. Another measure that can be used if an acceptable grazing arrangement can be devised is to utilize prescriptive grazing to stimulate new growth, control noxious weeds and increase native perennial plants in the area.

4.3.2 Red Rock River Mouth

The Red Rock River mouth area is rich in riparian communities and small mammal and migrating bird habitat. The area is used by anglers, hunters, and wildlife viewers. Improvements identified for this area include continuation of multi-use management of the area for hunting, fishing, and wildlife viewing access. Additionally, further noxious weed control should be implemented to suppress noxious weeds along the railroad grade and in open areas above the reservoir margin.

4.3.3 South Side Open Lands

Lone Tree to Hap Hawkins Campground - Improvements includes a continuation of management practices to exclude cattle and ORV uses and repair of fencing on an as-needed basis. Replace existing fencing in areas along antelope migration and use corridors to improve access for this species.

4.3.4 North Side Open Lands

On these lands from the dam to West Cameahwait, continue current management practices and add uniform signage to designated areas that are closed to ORV.

4.3.5 East Side Lands

Maintain existing Reclamation public access roads and access road easements. Install additional fencing to delineate easement lands near Buffalo Lodge, and remove obstructions in the road.

4.3.6 Beaverhead River Lands Below Dam

No new land management changes are proposed for the Beaverhead River Lands below the dam.

4.4 Cultural Resources Goals and Objectives

This section identifies goals, objectives and, where possible, actions applicable to guide future cultural resource management of Clark Canyon Reservoir lands. These following goals and objectives are intended to complement site-specific land area management prescriptions that follow.

4.4.1 Cultural Resource Goal

The primary goal for Cultural Resources Management is:

Protect cultural resources and provide educational/interpretive opportunities.

For this goal the objectives and general actions include:

- a. Ensure protection of sensitive cultural and paleontological resources in accordance with existing Federal regulations through consultation with the Shoshone-Bannock Tribes.
- b. Install signs at key access points to indicate damage of cultural resources is punishable by law, citing Archaeological Resources Protection Act and appropriate state codes.
- c. Cultural Resources Management Plan will be prepared.

As a further enhancement to cultural resources, an interpretive program would be developed in the Cultural Resources Plan that would highlight the history of the area and the important role this area played in the Lewis & Clark expedition. The program should include interpretive signs installed at public locations.

4.5 Operations, Maintenance, Enforcement

This section identifies goals, objectives and where possible, actions applicable to guide future operations, maintenance and enforcement of Clark Canyon Reservoir lands. These goals and objectives are intended to complement site-specific land area management prescription that follows.

The primary goal for operation maintenance and enforcement are:

Protect existing Reclamation natural resources at Clark Canyon Reservoir and Barretts Dam though a proactive program designed specifically to take into account recreational facilities operations, maintenance, and enforcement of existing rules and regulation that govern the recreation operation of the reservoir.

For this goal, the objectives and general actions include:

- a. Continue to provide a proactive maintenance program of existing and future recreation amenities that take into account public safety needs, accessibility needs, water quality protection, user demand, habitat protection, and cultural resource protection.
- b. Continue to provide an appropriate level of enforcement of existing rules and laws that govern the land-based activities, as well as, use of the facilities by the public. Continue to work cooperatively with other local agencies (e.g., State of Montana, Beaverhead County, U.S. Bureau of Land Management) to maximum the use of existing resources and funding to ensure adequate levels of enforcement are provided in order the balance public safety, resource protection, and water supply commitments.





Clark Canyon Draft Resource Management Plan for Clark Canyon Reservoir and Barretts Diversion Dam

Chapter 5 Implementation and Monitoring Program

5.0 Plan Implementation

Implementation of the RMP is primarily the responsibility of Reclamation. Because funding is uncertain, implementation of specific actions will require close coordination between Reclamation and the many other parties (stakeholders) interested in the management and use of Clark Canyon Reservoir and Barretts Diversion Dam.

Other factors that may influence the implementation of a particular action are based on whether the action: (1) is procedural or technical, such as preparing agreements or developing specific plans; (2) addresses public health and safety concerns; (3) is in compliance with existing laws, authorities, and regulations; (4) is required to prevent resource damage or protect wildlife species or habitats; or (5) requires large capital investments, such as trail or facility development.

Successful planning and coordination will be necessary to identify annual program priorities and will be essential in securing funding necessary to accommodate the goals and objectives of this RMP. To aid in planning for future needs and development at Clark Canyon Reservoir and Barretts Diversion Dam, Reclamation will serve as the lead agency to collaborate with appropriate stakeholders on a periodic basis to discuss issues, concerns, and solutions and identify funding sources.

Reclamation may revise or amend the RMP within the established 10-year planning period. During the implementation or monitoring phases of the RMP, Reclamation, other agencies, or the public may identify problems, deficiencies, or additional issues that should be addressed. Changes in the social, economic, physical, or environmental conditions may also necessitate changes to the RMP. Minor changes in data or materials that do not conflict with the established goals and objectives would be documented by Reclamation and would not require further public involvement and NEPA review. Changes that would modify one or more of the prescribed decisions and require major changes to the established goals and objectives would be documented by an amendment to the RMP and may require further public involvement and NEPA compliance. Reclamation will determine the level of public involvement and NEPA compliance.

Grazing has been successfully used elsewhere as a management tool and can be used at Clark Canyon Reservoir to reduce the build up of plant residue and litter by the trampling action of animal hooves and proper grazing. Before a grazing system is implemented, a grazing plan will be developed with the goal of improving wildlife habitat and enhancing desirable plant species while reducing noxious weeds. The grazing plan will be designed to prevent adverse impacts to the biological and physical resources of the area that could result if the areas are over grazed.



Plant residue build-up & dead willows near Horse Prairie Mouth

Prior to implementing grazing, a vegetative transect by range site or area will be set up along with photo points to establish a baseline condition. The plant species present will be documented, noting whether they are native or desirable to the site, the vigor, and percent bare ground. Permanent monitoring sites will be established where data can be collected every five to ten years. Change can be determined by documenting the vegetation through data collection and photos. The kind of

growing season and amount of precipitation during the year would be noted and whether it was below average, average or above.

Periodic monitoring will be conducted by Reclamation on an on-going basis throughout the 10-year RMP timeline. On-going evaluation of the monitoring program will allow Reclamation to make modifications in timing of improvements, timing of mitigation implementation, and changes in the RMP that are needed to take into account changing visitation needs or other changes in site conditions.



5.1 Alternatives Eliminated from Consideratio

White Top near boundary fence at Horse Prairie

Scientists considered and evaluated fire as a management tool to eliminate build up of plant residue and litter that is suppressing the vigor and health of plants. Also considered, was the use of fire to improve plant diversity for wildlife on some vegetative sites. Land management agencies and private landowners to maintain healthy plant communities have used prescribed burning. The sites mainly affected by the build up of plant residue and litter are the wet and semi-wet areas of the Horse Prairie Mouth, Red Rock River Mouth, and Beaverhead River Lands down stream of the Clark Canyon Dam. An effective prescriptive burn requires a prescriptive fire plan. Where fire can be used in a safe environment, monitoring and follow up are needed to assure the desired objectives are carried out and achieved. To determine a prescribed burn as a component of a management plan, one should assess resource condition, needs, and associated risks of a fire getting out of control. A successful fire plan would need to accomplish the desired objectives without adverse affects.

Reclamation has evaluated the resources needed and risks of a fire getting out of control and determined the associated risks are too great verses the possible benefit it would achieve and has eliminated this management tool as an option. There are several residence and public roads in the area including I-15 that create logistical problems and increase the potential risks of undertaking prescribed burns. Recent prescribed burns in New Mexico, Utah has verified the extensive damage and cost of a prescribed fire getting out of control.

Clark Canyon Draft Resource Management Plan for Clark Canyon Reservoir and Barretts Diversion Dam

Appendix A: Recreation Visitor Use Executive Summary

Recreation Visitor Use Executive Summary

This information summarizes the results of a recreation survey on Clark Canyon reservoir, conducted from May 24, 2003 to February 29, 2004. There were 448 visitors who completed the questionnaires at nine sites on the reservoir, one site immediately below the dam (Fishing Access), and one site down river (Barretts).

Demographic information

- The average age ranged from 40 years old at the Lewis and Clark site to 56 years old at Beaverhead.
- Groups were predominantly male between 63 percent to 100 percent males.
- 33% to 55% of visitors had attended college followed by 20%- 30% with a high school education.
- Professional and retired were the predominant occupations reported by visitors.
- Montana residents spent \$89.29 on average per trip while nonresident visitors spent \$330.23 on average per trip.
- Montanan's accounted for 50% of all visitors to Clark Canyon followed by visitors from Idaho (9%), Utah (8%) and California (5%).
- If visitors were from Montana, Beaverhead County residents used the area the most followed by Silver Bow, Ravalli, and Missoula counties.
- Visitors generally came as a family group or as a group of friends to Clark Canyon. Average group sizes at the various sites ranged from a high of 4.44/ group at the Cameahwait site to 2/group at the South Shore sites.

Trip Characteristics

- The majority of visitors were repeat visitors to the sites and 40%-85% had been to the site they were intercepted at more than 10 times. More than half of the visitors to five of the sites had been visiting that site for over 10 years.
- Most visitors (42% at Fishing Access to 100% at South Shore) did not spend a night.
- The most important reasons for choosing a site were easy to get to, good fishing, been there before, and scenic beauty. Fishing Access received the highest percent (70%) who indicated good fishing followed by 63% at Lone Tree. 63% of visitors at Overlook indicated the Lewis and Clark historic site as the reason for visiting that site.
- Sightseeing, walking, viewing wildlife, and fishing were the most popular activities. Ice fishing was engaged in the most at South Shore and Horse Prairie.

Satisfaction, Importance, and Facility Needs

- Visitors responded positively about their overall trip satisfaction.
- Cleanliness of the area and maintenance of the facilities received the most number of people indicating these conditions as important and each site received positive satisfaction levels related to cleanliness and maintenance.
- The condition of the campsite/picnic areas was important to most visitors and the visitors were satisfied with these conditions.

- The number of fish caught was very important to visitors at Fishing Access, South Shore, and Horse Prairie. Satisfaction level was the highest at Fishing Access with a positive response but a negative response in satisfaction of number of fish caught was reported at South Shore.
- The only agreement on facilities needed at each site was "none" indicating that visitors like the sites the way they are.
- When visitors wrote in suggestions for management and satisfaction, the predominant theme was that visitors encouraged management to keep the sites as a "no fee" area and that they liked the site as it currently exists.

Encounters, Conflicts, Crowding, and Place Attachment

- The number of encounters with people engaged in different types of activities was quite low at every site in the study. Because of this, conflicts at these sites were low to none.
- Fishing Access was the only site where more encounters occurred. 83% encountered bank anglers and 9% disliked seeing them; 77% encountered wade anglers and 9% disliked seeing them, and; 68% encountered boat anglers and 14% disliked seeing them.
- In general visitors were not at all crowded at Clark Canyon but one-third of the South Shore visitors felt slightly crowded.
- Visitors to two sites, Lone Tree and South Shore, appear to have a greater attachment to place than the other visitors at other sites. Attachment, however was not very strong at any site.

Clark Canyon Draft Resource Management Plan for Clark Canyon Reservoir and Barretts Diversion Dam

Chapter 6 List of Preparers

6.0 List of Preparers

Jan Anderson, Range Scientist/Resource Planner

Franson Noble Engineering 1276 South 820 East, Suite 100 American Fork, UT 84003

Jeff Baumberger, Project Lead

USBR- Great Plains Region, Montana Area Office 2900 4th Ave North, Suite 501 PO Box 30137 Billings, MT 59107-0137 Phone (406) 247-7330 jbaumberger@gp.usbr.gov

Rick Blaskovich, Environmental Specialist (retired)

USBR- Great Plains Region, Montana Area Office 2900 4th Ave North Suite 501 PO Box 30137 Billings, MT 59107-0137

Bob Delp, Recreation Resource Specialist

Argonaut Consulting LLC 8912 Sutter's Gold Dr. Sacramento CA 95826

Eric Franson, Hydrologist

Franson Noble Engineering 1276 South 820 East, Suite 100 American Fork, UT 84003

Jay W. Franson, P.E., Study Manager

Franson Noble Engineering 1276 South 820 East, Suite 100 American Fork, UT 84003

Lon Franson, Drafting Specialist

Franson Noble Engineering 1276 South 820 East, Suite 100 American Fork, UT 84003

Kirby Gilbert, Recreation Planner

Battelle Seattle Research Center 4500 Sand Point Way Seattle WA 98105-3949
CindaLee Hall, Secretary

Franson Noble Engineering 1276 South 820 East, Suite 100 American Fork, UT 84003

Kathy Kinsland, Environmental Scientist

Argonaut Ecological Consulting Inc. 7144 Fair Oaks Blvd, Suite 2 Carmichael, CA 95608

Steve Morehouse, Biological Sciences Technician

USBR – Great Plains Region, Montana Area Office Dillon Field Office 1100 Highway 41 Dillon, MT 59725 Phone (406) 683-6472 smorehouse@gp.usbr.gov

Mary Clare Schroeder, Wildlife Biologist

TetraTech FW Inc 12100 NE 195th St Bothell WA 98011

Nanette Shelton, Secretary

Franson Noble Engineering 1276 South 820 East, Suite 100 American Fork, UT 84003

Bill Vincent, Archeologist

USBR- Great Plains Region, Montana Area Office 2900 4th Ave North Suite 501 PO Box 30137 Billings, MT 59107-0137 **Clark Canyon Draft Resource Management Plan** for Clark Canyon Reservoir and Barretts Diversion Dam

Chapter 7 References Consulted

7.0 References Consulted

Alt, D. and Hyndman, D.W., 1986. *Roadside Geology of Montana*. Mountain Press Publishing Company. Missoula, Mt.

American Automobile Association, 1996. Highway Map of Idaho and Montana. Heathrow, Florida.

Anaconda Local Development Corporation, 1998. Accessed October 18, 2003. http://www.anacondamt.org/climate.asp

Anon, 1990. <u>The History of Beaverhead County, Vol. I</u>. Beaverhead County History Book Association. Dillon, Montana. Herff-Jones, Inc.

Beaverhead County Resource Use Committee. *Comprehensive Resource Use Plan for Beaverhead County, Montana*, July 26, 1999. Accessed October 30, 2003. http://www.beaverheadriver.com/bcruc/plan/aug99.html

Beaverhead County Weed Management Plan 2001. Beaverhead County Weed Board. Adopted in 1997, revised in 2001. Unpublished.

Beaverhead County, 2003. Data obtained from Beaverhead County home page Internet site at: <u>http://www.mtrdp.org/beaverhead/index.htm</u> on October 29, 2003. (Date of material and date of posting on Internet site not available.)

Beaverhead County, 2003. Information obtained from Beaverhead County website at: <u>http://www.mtrdp.org/beaverhead/index.htm</u>

Bureau of Economic Analysis, July 20, 2001. *Local Area Unemployment Statistics, Montana*. Accessed July 30, 2001. <u>ftp://ftp.bls.gov/pub/time.series/la/la.data.33.Montana</u>

Bureau of Economic Analysis. June 25, 2001a. *Regional Accounts Data, Local Area Personal Income, Montana, 1999.* Accessed July 26, 2001. <u>http://www.bea.gov/bea/regional/reis/action.cfm</u>

Bureau of Economic Analysis. June 25, 2001b. *Regional Accounts Data, Local Area Personal Income*. Beaverhead, Montana. 1999. Accessed July 26, 2001. http://www.bea.gov/bea/regional/reis/action.cfm

Deaver, K. and J. Morter. <u>Site Distribution in the Clark Canyon Reservoir Area, Southwest</u> <u>Montana: Final Report of a Class II Cultural Resource Inventory</u>. Unpublished report prepared by Professional Analysts, on file, Bureau of Reclamation, Montana Area Office, Billings, 1981.

Fenenga, F. "Appraisal of the Archeological and Paleontological Resources of the Jefferson River Basin, Montana." Unpublished report prepared by Missouri River Basin Survey, Smithsonian Institution, on file, Bureau of Reclamation, Montana Area Office, Billings, 1951. Garsjo, M.M., 2000. *Beaverhead County: Unit 604, Dillon Valley, Montana: Physiography, Drainage and Geology.* Written for USDA NRCS 2002, Draft Soil Survey of Dillon Valley, Part of Beaverhead County, MT 604. Unpublished. Dillon, MT NRCS Office.

Holt, D.C., 1956. *Investigation of Mineral Resources at Clark Canyon Reservoir Site, East Bench Unit.* Bureau of Mines, Spokane Field Office, Spokane, WA.

Howard, Janet L. 1995. *Lepus californicus*. U.S. Department of Agriculture, Forest Service, Rocky Mountain Research Station, Fire Sciences Laboratory (2003, March) Fire Effects Information System, [Online]. Available: <u>http://www.fs.fed.us/database/feis/</u> [Accessed 4/16/03].

Howard, Janet L. 1996. *Perognathus parvus*. In: U.S. Department of Agriculture, Forest Service, Rocky Mountain Research Station, Fire Sciences Laboratory (2003, March). Fire Effects Information System, [Online]. Available: <u>http://www.fs.fed.us/database/feis/</u> [Accessed 4/16/03].

Malone, M., and R. Roeder. Montana A History of Two Centuries. Published by University of Washington Press, Seattle, Washington. 1976.

McDonald, C. 2001. *Geologic and Hydrogeologic Investigation of the Clark Canyon Landslide, Southwestern Montana*. Montana Bureau of Mines and Geology. Open-File Report 442. December, 2001.

MDT, 2003. Montana Department of Transportation. August, 2003. <u>http://www.mdt.state.mt.us/</u>link to<u>http://www.byways.org/travel/state.html?CX_STATE=MT</u>

Minnesota IMPAN Group, Inc., June 2000. *IMPLAN Professional™ Version 2.0 Social Accounting & Impact Analysis Software, 2nd Edition*. Stillwater, Minnesota.

MNHP (Montana Natural Heritage Program) 2003b. *Federally Listed Endangered, Threatened, Proposed, and Candidate Species in Montana Counties*. Available on-line <u>http://www.fwp.state.mt.us/outsideurl.aspx?url=http://nhp.nris.state.mt.us/</u>. Last updated 9/2003. Accessed 11/10/2003.

MNHP (Montana Natural Heritage Program)2002. *Plant Species of Concern List and Field Guides*. Available on-line <u>http://nhp.nris.state.mt.us/plants/</u> Last updated 9/2/2002. Accessed 4/14/2003.

MNHP (Montana Natural Heritage Program) 2003. Letter listing species of concern in the vicinity of the Clark Canyon Reservoir. Helena, MT 3/14/2003

Montana Bureau of Land Management. *Narrative for Dillon Quadrangle, Dillon Resource Area Management Plan/EIS*. December 27, 1990. Accessed October 18, 2003. link to: <u>http://www.mt.blm.gov/oilgas/geostudies/dillon.pdf</u>

Montana Bureau of Mines and Geology 2002. Earthquake Studies Office web site. Last updated March 21, 2002. Accessed April 14, 2003. <u>http://mbmgquake.mtech.edu/</u>

Montana Department of Environmental Quality, 2003. *Montana 303(d) Information*. <u>http://www/deq.state.mt.us/ppa/mdm/303_d/303d_information.asp</u>

Montana Department of Environmental Quality. *Dillon Mining District, Beaverhead County*). Accessed October 30, 2003. <u>http://www.deq.state.mt.us/Rem/mwc/linkdocs/techdocs/12tech.asp</u>

Montana Department of Environmental Quality. *Montana Ambient Air Monitoring Network Review*. Accessed October 28, 2003. <u>http://deq.state.mt.us/ppa/mdm/air/nwHome.asp</u>

Moulton, G. The Journals of the Lewis & Clark Expedition Vol. 5. Published by University of Nebraska Press, Lincoln, Nebraska. 1997.

Oswald, R.A. 2002a. *Inventory and Survey of Fisheries in Lowland Lakes and Reservoirs of the Red Rock, Ruby, Beaverhead, and Big Hole River Drainages of Southwest Montana*. Montana Department of Fish, Wildlife & Parks. Bozeman, Montana. June, 2002.

Oswald, R.A. 2002b. *Inventory and Survey of Selected Stream Fisheries of the Red Rock, Ruby, and Beaverhead River Drainages*. Montana Department of Fish, Wildlife & Parks. Bozeman, Montana. June, 2002.

Reclamation, 2002. *Recreation Management Condition Assessment* – Clark Canyon Reservoir and Barretts Diversion Dam. United States Department of the Interior, Bureau of Reclamation, Great Plains Region, Montana Area Office. August, 2002.

Snyder, S. A. 1993. *Haliaeetus leucocephalus*. In: U.S. Department of Agriculture, Forest Service, Rocky Mountain Research Station, Fire Sciences Laboratory (2003, March). Fire Effects Information System, [Online]. Available: <u>http://www.fs.fed.us/database/feis/</u> [Accessed 4/16/03].

Stickney, M.C. 1995. *Seismic Sources Zones in Southwest Montana*. Montana Bureau of Mines and Geology, Montanan Tech of the University of Montana. USGS Award No. 1434-95-G-2628. Accessed October 26, 2003. link to: http://erp-web.er.usgs.gov/reports/abstract/1995/ni/g2628.html

Stickney, M.C. 1995. *Seismic Sources Zones in Southwest Montana*. Montana Bureau of Mines and Geology, Montanan Tech of the University of Montana. USGS Award No.; 1434-95-G-2628. Available on-line:

http://erp-web.er.usgs.gov/reports/abstract/1995/ni/g2628.htm

Tesky, Julie L. 1994. *Brachylagus idahoensis*. U.S. Department of Agriculture, Forest Service, Rocky Mountain Research Station, Fire Sciences Laboratory (2003, April). Fire Effects Information System, [Online]. Available: <u>http://www.fs.fed.us/database/feis/</u> [Accessed 4/18/03].

Tesky, Julie L. 1994. *Buteo regalis*. In: U.S. Department of Agriculture, Forest Service, Rocky Mountain Research Station, Fire Sciences Laboratory (2003, March). Fire Effects Information System, [Online]. Available: <u>http://www.fs.fed.us/database/feis/</u> [Accessed 4/16/03].

Texas Parks and Wildlife. Threatened and Endangered Species. *Eagles*. <u>http://www.tpwd.state.tx.us/nature/endang/animals/birds/baldeagl/htm</u>

U.S. Census Bureau, May 2001a. *Profiles of General Demographic Characteristics 2000. 2000 Census of Population and Housing, Idaho.* Accessed July 25, 2001. <u>http://www.census.gov/prod/cen2000/dp1/2kh16.pdf</u>

U.S. Census Bureau, May 2001b. *Profiles of General Demographic Characteristics 2000.* 2000. Census of Population and Housing, Montana. Accessed July 25, 2001. <u>http://www.census.gov/prod/cen2000/dp1/2kh30.pdf</u>

U.S. Department of Agriculture, National Resources Conservation Service, 2002. *Draft Soil Survey of Dillon Valley, Part of Beaverhead County, MT 604*. Unpublished. Dillon, MT NRCS Office.

U.S. Environmental Protection Agency. Accessed October 26, 2003. http://www.epa.gov/aqspubl1/select.html

U.S. Fish and Wildlife Service, 1956. *A Detailed Report on Fish and Wildlife Resources in Relation to the Water Development Plan for the East Bench Unit.* Three Forks Division. Missouri River Basin Project, MT. June, 1956.

US Fish and Wildlife Service. Division of endangered species. Volume 60, Number 133, pp 36000-36010. July 12, 1995. <u>http://endangered.fws.gov/fr95580.html</u>

US Department of the Interior 1965. *Reservoir Management Plan; Clark Canyon reservoir and Barretts Diversion Dam.* Bureau of Reclamation. Billings, MT. 1965.

US Department of the Interior 2002. *Recreation Management Condition assessment; Clark Canyon Reservoir and Barretts Diversion Dam.* US DOI Bureau of Reclamation, Montana Area Office. August, 2002.

US Fish and Wildlife 2002. Letter listing Federally endangered and threatened species potentially of occurring in the vicinity of Clark Canyon Reservoir and Barretts Diversion Dam in Beaverhead County. File No. M.04 BR(1). 12/13/2002. Helena, MT.

USDA NASS 2002. *Montana Agricultural Statistical Service*. Climataological Data. Available online: <u>http://www.nass.usda.gov/mt/cropwthr/climate.htm#top</u>

USDA NRCS 2002. *Draft Soil Survey of Dillon Valley, Part of Beaverhead County, MT 604*. Unpublished. Dillon, MT NRCS office.

Western Regional Climate Center. Climate data for Dillon, MT for the period of 1895 to 2003. Accessed October 26, 2003. link to: <u>http://www.wrcc.dri.edu/CLIMATEDATA.html</u>

WRCC (Western Region Climate Center). *Division of Atmospheric Sciences Desert Research Institute*. Reno, NV. Available on-line: <u>http://www.wrcc.dri.edu</u> Accessed 4/14/2003